DISCUSSION PAPER SERIES

DP13957

HOME BIAS IN HUMANITARIAN AID: THE ROLE OF REGIONAL FAVORITISM IN THE ALLOCATION OF INTERNATIONAL DISASTER RELIEF

Christian Bommer, Axel Dreher and Marcello Perez-Alvarez

DEVELOPMENT ECONOMICS

PUBLIC ECONOMICS



HOME BIAS IN HUMANITARIAN AID: THE ROLE OF REGIONAL FAVORITISM IN THE ALLOCATION OF INTERNATIONAL DISASTER RELIEF

Christian Bommer, Axel Dreher and Marcello Perez-Alvarez

Discussion Paper DP13957 Published 27 August 2019 Submitted 22 August 2019

Centre for Economic Policy Research 33 Great Sutton Street, London EC1V 0DX, UK Tel: +44 (0)20 7183 8801 www.cepr.org

This Discussion Paper is issued under the auspices of the Centre's research programmes:

- Development Economics
- Public Economics

Any opinions expressed here are those of the author(s) and not those of the Centre for Economic Policy Research. Research disseminated by CEPR may include views on policy, but the Centre itself takes no institutional policy positions.

The Centre for Economic Policy Research was established in 1983 as an educational charity, to promote independent analysis and public discussion of open economies and the relations among them. It is pluralist and non-partisan, bringing economic research to bear on the analysis of medium- and long-run policy questions.

These Discussion Papers often represent preliminary or incomplete work, circulated to encourage discussion and comment. Citation and use of such a paper should take account of its provisional character.

Copyright: Christian Bommer, Axel Dreher and Marcello Perez-Alvarez

HOME BIAS IN HUMANITARIAN AID: THE ROLE OF REGIONAL FAVORITISM IN THE ALLOCATION OF INTERNATIONAL DISASTER RELIEF

Abstract

Natural disasters represent a major challenge for human welfare across the globe. Given the prominent role of international humanitarian aid in alleviating human suffering, the investigation of its determinants is of paramount importance. While existing studies show its allocation to be influenced by donors' foreign policy considerations, domestic political factors within recipient countries have not been systematically explored. This paper addresses this important research gap by investigating whether regional favoritism shapes humanitarian aid flows. Using a rich and unique dataset derived from reports of the Office of US Foreign Disaster Assistance (OFDA), we show that substantially larger amounts of aid are disbursed when natural disasters hit the birth region of the recipient countries' political leader. While we find no evidence that US commercial or political interests affect the size of this home bias, the bias is stronger in countries with a weaker bureaucracy and governance, suggesting the absence of effective safeguards in the allocation of aid.

JEL Classification: H84

Keywords: Humanitarian Aid, Natural Disasters, Regional Favoritism, Birth Regions

Christian Bommer - christian.bommer@wiwi.uni-goettingen.de University of Goettingen

Axel Dreher - mail@axel-dreher.de University of Heidelberg and CEPR

Marcello Perez-Alvarez - marcello.perez@wiwi.uni-goettingen.de University of Goettingen

Acknowledgements

We thank Cooper Drury and Thomas Eisensee for generously sharing their OFDA data. We further thank Stephan Klasen, Sven Kunze, Sebastian Vollmer and conference participants at ETH Zürich (AEL 2018), MIT (IPES 2018) and Oxford University (CSAE 2019) for helpful comments, as well as Patrick Betz, Theresa Geyer, Adrian Monninger, Friederike Schilling, and Wangzhen Xia for excellent research assistance.

Home bias in humanitarian aid: The role of regional favoritism in the allocation of international disaster relief

Christian Bommer^a (University of Goettingen, Heidelberg University) Axel Dreher^b (Heidelberg University, University of Goettingen, KOF, CEPR, CESifo) Marcello Perez-Alvarez^c (University of Goettingen)

Abstract

Natural disasters represent a major challenge for human welfare across the globe. Given the prominent role of international humanitarian aid in alleviating human suffering, the investigation of its determinants is of paramount importance. While existing studies show its allocation to be influenced by donors' foreign policy considerations, domestic political factors within recipient countries have not been systematically explored. This paper addresses this important research gap by investigating whether regional favoritism shapes humanitarian aid flows. Using a rich and unique dataset derived from reports of the Office of US Foreign Disaster Assistance (OFDA), we show that substantially larger amounts of aid are disbursed when natural disasters hit the birth region of the recipient countries' political leader. While we find no evidence that US commercial or political interests affect the size of this home bias, the bias is stronger in countries with a weaker bureaucracy and governance, suggesting the absence of effective safeguards in the allocation of aid.

Keywords: humanitarian aid, natural disasters, regional favoritism, birth regions

Acknowledgments: We thank Cooper Drury and Thomas Eisensee for generously sharing their OFDA data. We further thank Stephan Klasen, Sven Kunze, Sebastian Vollmer and conference participants at ETH Zürich (AEL 2018), MIT (IPES 2018) and Oxford University (CSAE 2019) for helpful comments, as well as Patrick Betz, Theresa Geyer, Adrian Monninger, Friederike Schilling, and Wangzhen Xia for excellent research assistance.

^a University of Goettingen, Faculty of Economic Sciences and Centre for Modern Indian Studies, Waldweg 26, 37073 Göttingen, Germany; Email: christian.bommer@wiwi.uni-goettingen.de

^b Heidelberg University, Alfred-Weber-Institute for Economics, Bergheimer Strasse 58, 69115 Heidelberg, Germany; Email: mail@axel-dreher.de

^c University of Goettingen, Faculty of Economic Sciences, Platz der Goettinger Sieben 5, 37073 Göttingen, Germany; Email: marcello.perez@wiwi.uni-goettingen.de

1. Introduction

The principle of impartiality in the allocation of humanitarian aid is firmly established in international law (Persson 2004). In spite of this, anecdotal evidence suggesting that politically important subnational regions receive favorable treatment is easy to find. According to policy reports by the Red Cross and Red Crescent Societies (Klynman et al., 2007) and the International Dalit Solidarity Network (2013), power relations at the community level within recipient countries distort the allocation of humanitarian aid. In this paper, we provide the first systematic investigation of whether and to what extent humanitarian aid is indeed impartial with respect to recipient country politics. We focus on the birth regions of recipient country leaders and investigate whether they are more likely to receive (larger) support when being hit by exogenous rapid-onset natural disasters.

The importance of national leaders' birth regions for the allocation of funds under their control has been demonstrated in previous work, most notably in Hodler and Raschky (2014). Investigating one potential channel, Dreher et al. (2019) show that recipient leaders channel foreign aid to their birth regions to the extent that the donor does not put strings on how these funds are allocated, but not otherwise. We thus consider the focus on birth regions to be a suitable test of the impartiality principle in the allocation of humanitarian aid. Given the direct connection of such assistance with humanitarian suffering, the examination of such political economy factors is of paramount importance.¹

We examine whether recipient country leaders can channel humanitarian aid in line with their personal interests – with the potential to influence domestic political equilibria – and whether and to what extent commercial and political relations with the donor facilitate their abuse. Specifically, we investigate the allocation of humanitarian aid from the United

¹ Natural disasters constitute a major challenge for human welfare. In the 1964-2017 period, they have reportedly killed more than five million people across the globe (Guha-Sapir et al., 2018). Furthermore, natural disasters exert a substantial negative effect on economic growth (Felbermayr and Gröschl, 2014) and on the human capital (Caruso, 2017) and income (Karbownik and Wray, 2019) of subsequent generations. As climate scientists predict a substantial increase in both the frequency and intensity of natural disasters in the near future, this type of aid is likely to further gain in importance for human welfare. What is more, climate-related risks are projected to be disproportionately concentrated in already vulnerable countries with low response capacities (IPCC, 2018).

States for 6,228 rapid-onset natural disasters that have hit 50 countries over the 1964-2017 period. We derive these rich and unique data on disaster relief from annual reports issued by the Office of US Foreign Disaster Assistance (OFDA) – the US agency responsible for providing disaster relief overseas. OFDA responds to an average of 65 disasters in more than 50 countries per year (USAID, 2018a); the United States have been by far the biggest donor of humanitarian aid. In the 1972-2017 period, they financed more than 40% of the humanitarian assistance of the countries that report to the OECD's Development Assistance Committee (DAC) (OECD, 2019).

We employ three strategies to address endogeneity and identify the causal effects of leaders' birth regions on the allocation of humanitarian aid.² First, we control for a range of observable characteristics of the affected subnational area. This type of specification allows us to adjust for the most obvious sources of confounding. However, unobserved omitted variables could potentially still bias estimates. In a second step, we therefore include disaster-area fixed effects, limiting the analysis to identical areas that have been hit by multiple disasters, while having experienced changes in their birth region status over time. In this restrictive setting all time-invariant unobserved heterogeneity is controlled for, allowing us to further increase the internal validity of our estimates.

As a third empirical approach, we run placebo regressions that test whether disasters hitting regions that were the birth region in the previous or subsequent year of the disaster, but not during the time of the disaster itself, receive similar treatment compared to disasters hitting contemporaneous birth regions. In comparing these points in time, any unobserved characteristics of birth regions that do not vary over a very short period are thus accounted for. We have no reason to assume that exogenous disasters should be more likely to receive funding in case they hit the birth region of a national leader, compared to disasters that hit the same region in the years directly before or after the

² While the timing as to when a rapid-onset natural disaster hits a particular sub-national region is random, decisions on aid allocation might be endogenous. Sub-national regions connected to the government by virtue of being the political leader's birth region might differ from other regions in ways that are correlated with the need for aid. For instance, it seems plausible that regions with political ties to the government are richer and better protected against the risks arising from natural disasters compared to areas populated by weaker groups. In such a case, our estimate of how regional favoritism affects the allocation of aid could be biased.

leader assumes power. Comparing these years thus allows us to derive a causal estimate of the importance of birth region favoritism in a scenario of urgent humanitarian need.

Our results suggest that birth region-related favoritism exerts a strong and robust influence on humanitarian aid, increasing the *amounts* of US-provided disaster relief by 45% to 85% in our main specifications. In contrast, we do not observe any systematic effects on the *probability* of receiving US-provided disaster relief, which can be plausibly explained by the structure of OFDA's decision-making process. We do not find evidence that the United States' political or commercial interests in a country hit by disaster affect the size of the home bias. To the contrary, recipient-country characteristics in which leaders should find it easier to misappropriate funds – such as clientelism prevailing in public spending or low bureaucratic quality – explain a substantial share of it. What is more, providing a novel and innovative dataset on ethnic power relations, we show that the observed favoritism is not explained by ethnic ties between political leaders and the population in disasteraffected areas, but constitutes a self-sustained, independent dimension of favoritism within recipient countries.

With this paper, we mainly contribute to two strands of literature. First, and most directly, our research connects to a number of studies that investigate the determinants of disaster aid allocation across countries. According to these studies, donor political interests influence humanitarian aid (Ball and Johnson, 1996; Drury et al., 2005; Eisensee and Strömberg, 2007; Strömberg, 2007; Fink and Redaelli, 2011; Fuchs and Klann, 2012; Raschky and Schwindt, 2012; Annen and Strickland, 2017). While some of these previous studies also used individual disasters as unit of observation (rather than country-years), the political economy within *recipient* countries has largely been ignored.³

³ There are two exceptions, both focusing on individual countries rather than a broader sample. One focuses on the head of government's electoral incentives, such as re-election concerns or providing support to the governing party at the national or regional level (Jayne et al., 2002; Francken et al., 2012, Kunze and Schneider, 2019; Eichenauer et al., 2019). The second investigates discrimination against individual victims of disasters. This literature shows that the probability to receive aid depends on gender, race, income, and education, among others (Broussard et al., 2014; Bolin and Kurtz, 2018).

Second, this paper contributes to the literature investigating regional favoritism. Hodler and Raschky (2014) and Dreher et al. (2019) show that country leaders' birth regions experience higher economic growth and receive more Chinese foreign aid, respectively. This literature provides the analytic background for our study. It shows that leaders of countries around the world divert resources to their birth regions, either for altruistic purposes, or political ones. However, given that disaster relief is primarily triggered by large, exogenous shocks with potentially grave humanitarian consequences, such political-economy considerations would even be more alarming, and stand in direct contrast to international law.

More broadly, our results also relate to the aid allocation literature at large. Much of this literature investigates how donor political interests affect the allocation of aid at the country level (Alesina and Dollar, 2000; Kuziemko and Werker, 2006; Hoeffler and Outram, 2011; Vreeland and Dreher, 2014). Only recently, this literature has begun to investigate political motives in the allocation of aid at the sub-national level. However, due to limited data availability, previous work has focused on either single recipient countries, multilateral aid, or the allocation of Chinese development finance.⁴ With this analysis, we are thus the first to investigate the effect of regional favoritism in the allocation of a Western bilateral donor for a large number of recipient countries and thus extend the literature in an important dimension. What is more, our focus on exogenous variation in recipient need caused by natural disasters facilitates the identification of causal effects compared to studies focusing on the allocation of aid more broadly.

Finally, we also contribute to the literature on ethnic power relations by providing novel data on the sub-national location of ethnic groups, which we use to separate favoritism towards ethnic regions from those towards birth regions. Unlike previous studies (e.g., De Luca et al. 2018, Anaxagorou et al., 2019), we identify the ethnic composition of sub-national populations based on census and survey data. This is an improvement over previous research, which is mostly based on less precise historical ethno-linguistic or expert-based maps such as *Ethnologue* data (Gordon, 2005), the *Geo-referencing of*

⁴ See Francken et al. (2012), Dionne et al. (2013), Jablonski (2014), Masaki (2018), Nunnenkamp et al. (2017), Brazys et al. (2017), Anaxagorou et al. (2019) and Dreher et al. (2019).

Ethnic Regions (*GREG*) data (Weidmann et al., 2010) or *GeoEPR* – a geocoded version of the *Ethnic Power Relations* (*EPR*) dataset (Wucherpfennig et al., 2011).

We proceed as follows. Section 2 introduces our main data sources – on humanitarian aid, natural disasters, leaders' birth regions – as well as our control variables, and discusses the methods used to construct our measures. Section 3 discusses descriptive statistics, while Section 4 explains our method of estimation. We show the main results in Section 5 and our analyses of whether US-commercial and political interests drive the effect in Section 6. Section 7 presents extensions to the main analyses, while the final Section 8 discusses the policy implications of our research.

2. Data sources and variables

Disaster Aid

Three main data sources have previously been used to study the effect of natural disasters on aid. One group of papers relies on data that donor governments report to the OECD's DAC, which includes entries for emergency relief and food aid, among others. While these data have the advantage of being easily available for the major Western donors organized in the DAC, using them comes at a cost. As DAC data do not exclusively focus on disaster relief, but also on crisis prevention, and emergencies other than disasters, such aid cannot directly be attributed to individual disasters (Fink and Redaelli, 2011). Given that it is thus not possible to attribute the aid flows to subnational regions in aid-receiving countries, DAC data are not suitable to address our research questions.

A second set of papers relies on data provided by the Financial Tracking Service (FTS) managed by the UN Office for the Coordination of Humanitarian Affairs (OCHA). One of its main advantages consists of its wider coverage of donor countries. These data cover nearly all donor countries in the world rather than just the set of mainly Western donors that report to the DAC. Contrary to the DAC data – which are at the recipient-year level – the FTS provides aid information for disaster appeals. These data come, however, with the disadvantage that reporting to FTS is voluntary, potentially giving rise to

underreporting (Harmer and Cotterrell, 2005; Raschky and Schwindt, 2012). Moreover, parts of the aid reported there cannot be attributed to specific disasters. While previous work making use of FTS data was restricted to the 1992-2004 period (Fink and Radaelli, 2011), the share of contributions not assigned to a specific disaster is substantially higher in more recent years, creating the risk of measurement error for our analysis.⁵ Finally, the lack of a common standard by which individual donors report their aid contributions makes it difficult to compare aid across donors. For example, overvalued in-kind contributions could lead to over-reporting of aid (Harmer and Cotterrell, 2005).

The third group of papers focuses on the United States as donor exclusively, using data from the US Office for Foreign Disaster Assistance (OFDA). OFDA is part of the US Agency for International Development (USAID) and is the lead agency responsible for international disaster relief (Drury et al., 2005; Eisensee and Strömberg, 2007; Fink and Radaelli, 2011; Margesson, 2013; Kevlihan et al., 2014).⁶ The United States are by far the largest provider of disaster relief, both in terms of the number of supported disasters and regarding the amounts of aid provided. For the 1972-2017 period, the United States financed more than 40% of the humanitarian assistance provided by countries of the DAC. (OECD, 2019). The key advantage of OFDA data compared to the previously mentioned data sources consists of the agency providing detailed and complete annual reports for each fiscal year since 1964, allowing us to match individual contributions to specific disasters without ambiguity. This rather long time frame combined with the reliability and completeness of aid data makes OFDA the most suitable data source for the research question at hand.

We extract aid flows from OFDA's annual reports for the fiscal years 1990 to 2017 and transform them to constant 2017 US Dollars (OFDA, 1990-2017). For the 1964-1989 period, we use data provided to us by Cooper Drury who previously conducted research

⁵ According to personal correspondence with FTS staff, one reason for the increase in unassigned contributions is a lack of resources to properly categorize all disasters (email from August 10th, 2017).

⁶ The USAID (2005: 10) defines disaster aid as "[i]mmediate, life sustaining assistance provided to disaster victims." It is given based "upon the written determination that a disaster exists in the host country which meets three criteria: it is of a magnitude with which the affected community cannot cope; recognized representatives of the affected population desire the assistance; and it is in the USG's [United States Government's] interests to respond" (USAID, 2005: 5).

based on data from OFDA's annual reports (Drury et al., 2005).⁷ Besides OFDA, there are other US agencies that allocate disaster assistance, most notably the Office of Food for Peace (also part of USAID) and the US Department of Defense (Margesson, 2013). Thanks to its chief role for US disaster assistance, OFDA collaborates closely with these agencies and provides approximate numbers for their aid contributions for the sub-period 1964 to 2004.⁸ While our main analysis focuses on the full period (1964-2017), we show in a test for robustness that our key results extend to the aid provided by the full range of US agencies using the 1964 to 2004 sub-period.

Natural Disasters

We take data on natural disasters from the EM-DAT international disaster database (Guha-Sapir et al., 2018), which was assembled by the University of Louvain's *Centre of Research on the Epidemiology of Disasters* (*CRED*). This is the most comprehensive list of disasters available, comprising more than 22,000 disasters from the year 1900 to the present. EM-DAT includes disasters that meet at least one of the following criteria: At least ten people are reported as killed, at least one hundred people are reported to be affected, a state of emergency has been declared, or a call for international assistance has been issued. They provide disaster-specific information on the type of disaster, the number of people killed, missing and presumed dead, and the number of people affected (EM-DAT, 2018b).⁹ Conveniently, EM-DAT also includes the sub-national location of the disaster for the first administrative levels or lower.¹⁰

For our analysis, we include the 50 countries that are most frequently affected by rapidonset disasters (floods, storms, earthquakes, epidemics, landslides, extreme temperatures, volcanic activity, wildfires, dry mass movements and insect infestations)

⁷ All annual reports (with the exception of the 1974-1982 period) are available for download from the USAID Development Experience Clearinghouse (USAID, 2018b).

⁸ In this time period, OFDA aid flows represent on average 76% of all US disaster relief.

⁹ EM-DAT draws from a number of sources, including UN agencies, non-governmental organizations, insurance companies, research institutes and press agencies (EM-DAT, 2018a).

¹⁰ Subnational regions at the first administrative level (ADM1) typically are departments, provinces or states; regions at the second administrative level (ADM2) include districts or municipalities, among others. Whenever a change in the admin boundaries occurred over time, we consider the modern counterparts of reported administrative areas. Note that out of 7,318 disasters, 880 disasters (of which only 60 received OFDA aid) are not considered in the analysis due to imprecise location information. Moreover, disasters in Vietnam during the Vietnam War are excluded.

over the 1964-2017 period.¹¹ We follow Fink and Redaelli (2011) in limiting the scope of the analysis to rapid-onset disasters, whose timing is rather random and therefore occur unexpectedly. By contrast, the timing and particularly the duration of slowly evolving catastrophes such as droughts, famines or complex emergencies might introduce endogeneity concerns.¹² We merge the EM-DAT list of disasters with disaster-specific aid data from OFDA reports based on the timing and location of disasters. EM-DAT disasters not mentioned in the OFDA reports are assumed to not have received aid.¹³

Leaders' birth regions

To identify birth regions of the countries' political leaders, we make use of the latest version of the Archigos database of leaders such as presidents, prime ministers, or religious leaders, depending on the political system (Goemans et al., 2009; Archigos, 2018). We complement these data with birthplace information acquired through online search. While leader birth regions are often also available at the ADM2 level, we focus on ADM1 birth regions, as EM-DAT does not always provide a comprehensive list of affected ADM2 areas. Our explanatory variable of interest is then constructed as a binary indicator equal to one if any of the disaster-specific locations listed by EM-DAT was the birth region of the political leader at the time of the disaster, and zero otherwise.¹⁴

¹¹ Given that we focus on within-country variation in disaster locations, increasing the sample beyond the 50 most frequently affected countries would hardly increase the degrees of freedom. Moreover, the following advanced economies with a high number of natural disasters, which rarely appear in OFDA reports, were not included in the sample: Australia, Canada, France, Hong Kong, Italy, Japan, New Zealand, Russia, South Korea, Spain, Taiwan, and United Kingdom. We exclude Algeria due to a lack of variation in the birth region status of disaster-affected areas and Somalia due to the uncertain governmental control over national territory.

¹² Including disasters that are driven by such dynamics would substantially increase the complexity of the analysis and its interpretation as the United States may adjust its aid depending on how a disaster evolved. For the same reason, we do not include technical disasters.

¹³ We exclude observations for which matching was not possible due to ambiguous information on timing, and location and due to ambiguous information on aid flows. As a consequence, 79 disasters remain unmatched and hence drop out from the sample.

¹⁴ For most disasters reported by EM-DAT, start dates and end dates are identical or very close to each other. In the uncommon event where start and end dates differ and leaders change in between, we code the binary indicator as equal to one if any of the leaders' birth regions was hit. We use the same approach for the rare case that only the month and year of the disaster are reported and a leader change happens to occur in the respective month.

Control variables

To address the concern that regions affected by natural disasters may differ from other regions in ways that are correlated with birth region status and aid flows, we construct a range of control variables reflecting local area characteristics. In particular, we suspect that birth regions are richer, more populated (affecting the economic damage and number of casualties created by disasters) and easier to access (facilitating disaster relief). For the construction of control variables for population density, nighttime light intensity, barren land and ruggedness, we match the disaster locations indicated by EM-DAT with grid maps and calculate zonal statistics using a Geographic Information Systems software. For the case of multi-location disasters, we weight locations by their area.¹⁵

As a measure for population concentration, we use the above procedure to construct a continuous variable for average population density per square kilometer in the year 2015.¹⁶ Moreover, we control for the population size in major cities of affected areas for the entire time period of analysis, based on data from the UN World Urbanization Prospects (UN, 2018). We lag values by one year, rescaled to ten million inhabitants.¹⁷

For economic activity, we use average annual cloud-free night time light intensity maps in combination with population figures to construct a measure of average night time light intensity per capita. This variable aims to capture economic activity conditional on population density. For that purpose, we use data from the National Oceanic and Atmospheric Administration (NOAA, n.d.). As night time lights are highly volatile across years, we exploit the availability of maps for multiple years (1992 to 2013) and take averages across time before taking averages across administrative regions. This approach brings the advantage of reducing the influence of very cloudy years, in which satellites can only measure night time light emissions during a limited number of days.¹⁸

¹⁵ To increase precision, we exploit information at the ADM2-level whenever available.

¹⁶ We rescale the variable to 1,000 people per square kilometer. For the vast majority of countries, we use data from the WorldPop project (Tatem, 2017). Where these are unavailable, we use SEDAC (CIESIN, 2017).

¹⁷ The cities included in the dataset are those with more than 300,000 inhabitants in the year 2018.

¹⁸ The fact that we take averages across time brings the disadvantage that we cannot use lagged night time light intensity which would limit endogeneity concerns. However, the required satellite data are unavailable for years before 1992 and we consider measurement error a more severe problem. Moreover, results

Further variables that control for the economic or strategic relevance of the affected areas are the percentage of barren land, the number of ports, the number of nuclear plants and a binary indicator for disasters hitting the capital city. The barren land variable is based on the GlobCover 2009 grid map, which captures the share of land classified as bare land in the disaster-affected areas (Arino et al., 2012). Data for the number of large ports and nuclear plants comes from the World Port Index of the US National Geospatial Intelligence Agency (NGIA, 2017) and the World Nuclear Association (WNA, 2017), respectively.¹⁹

To proxy for accessibility, we measure ruggedness with the so-called Terrain Ruggedness Index, which was initially developed by Riley et al. (1999) and constitutes a fine-grained measure for average differences in elevation per 30 arc seconds grid cell. We use preconstructed grid cell-level data by Nunn and Puga (2012) and take averages across grid cells within disaster areas. Similar to them, we scale the index such that it represents average elevation differences in hundreds of meters.

3. Data description

Table 1 shows descriptive statistics by disaster type. The 50 countries included in our sample were hit by 6,228 natural disasters, 13.4% of which received disaster relief from OFDA. On average, 360 persons were reported dead and the average disaster with positive aid flows received approximately 1.78 million 2017 US dollars. Funding shows high variation as indicated by the large standard deviations in parentheses. In rare cases, OFDA donated aid to multiple disasters at once. We treat these cases as one event and assign the disaster type "multiple" if these disasters were of different types.

⁽available on request) remain close to identical when night light intensity and/or population density are dropped from the analysis.

¹⁹ We use one-year lags for the number of nuclear plants. The number of ports does not vary over time in our sample period.

Disaster type	Frequency	Average number of casualties (SD)	Percentage funded	Average funding in 1,000s, 2017 US dollars (SD)
Earthquake	761	1,350.1 (13,058.8)	18.4	4,850.8 (28,916.2)
Epidemic	608	194.0 (600.6)	5.9	865.8 (3,666.2)
Extreme temperature	186	134.0 (315.5)	1.6	1,732.6 (2,798.0)
Flood	2554	79.7 (654.5)	13.9	1,047.5 (12,695.0)
Insect infestation	16	0.0 (0.0)	37.5	537.3 (551.8)
Landslide	452	70.6 (185.3)	7.1	137.1 (215.3)
Mass movement (dry)	24	77.3 (86.8)	12.5	46.7 (7.0)
Storm	1371	583.9 (9,749.8)	13.8	1,665.4 (5,220.5)
Volcanic activity	134	183.2 (1,883.0)	22.4	905.7 (1,947.9)
Wildfire	98	8.0 (22.6)	18.4	485.2 (996.6)
Multiple	24	367.3 (397.9)	100.0	1,641.8 (3,048.3)
All disasters	6,228	360.0 (6,495.9)	13.4	1,776.5 (14,720.2)

Table 1. Disaster impact and emergency relief, 50 countries, 1964-2017

Notes: Standard deviations in parentheses. Average funding includes only aid provided by OFDA (rather than all US government agencies). We exclude disasters without any funding from this average.

Analyzing the table by disaster type, important differences become visible. First, certain disasters are substantially more frequent than others. For instance, more than half of the sample comprises of storms and flood events, while insect infestations are rare. Disasters belonging to the categories of insect infestation, volcanic activity, earthquake and wildfire are more likely to receive funding. Among those disasters with funding, earthquakes, extreme temperatures and storms receive the larger aid amounts on average. It should be noted however that there is a large variation in the granted aid amounts within most disaster types.

Country	Disasters	Average number of casualties	Percentage hit leader's birth region	Percentage funded	Average funding in 1,000s (2017 USD)
Afghanistan	150	137.7	7.3	5.3	509.9
Angola	55	109.3	41.8	9.1	293.6
Argentina	93	11.3	31.2	12.9	169.8
Bangladesh	223	802.4	29.1	4.5	2,413.4
Bolivia	67	25.1	38.8	34.3	232.2
Brazil	175	60.5	17.1	13.7	147.3
Chile	77	36.8	35.1	16.9	1,453.0
China	748	586.0	9.5	4.1	311.9
Colombia	143	212.3	19.6	14.0	771.2
Costa Rica	50	9.3	24.0	52.0	392.3
Cuba	49	4.3	34.7	8.2	519.3
DR Congo	118	89.8	16.1	8.5	923.3
Dom. Rep.	57	31.5	33.3	21.1	816.6
Ecuador	77	108.3	28.6	28.6	621.4
El Salvador	37	94.2	45.9	32.4	2,196.3
Ethiopia	59	44.5	27.1	10.2	173.7
Greece	69	9.0	24.6	13.0	504.9
Guatemala	77	339.2	35.1	16.9	634.7
Haiti	93	2,575.5	47.3	23.7	19,524.0
Honduras	52	232.0	26.9	28.8	687.3
India	535	326.8	8.2	7.9	1,081.4
Indonesia	415	72.3	10.6	12.3	474.6
Iran	159	744.1	6.9	5.0	2,076.1
Kenya	81	56.5	21.0	4.9	1,914.1
Madagascar	56	80.8	44.6	42.9	580.6
Malawi	40	32.5	22.5	22.5	366.4
Malaysia	60	15.6	20.0	11.7	415.8
Mexico	208	77.8	4.8	11.1	746.4
Mozambique	70	59.7	38.6	22.9	1,922.3
Myanmar	55	2,569.9	20.0	21.8	3,638.3
Nepal	83	231.3	38.6	16.9	2,676.2
Nicaragua	52	19.5	23.1	40.4	304.0
Niger	61	149.6	39.3	18.0	156.4
Nigeria	108	249.3	10.2	4.6	147.2
Pakistan	185	2,418.6	17.8	16.2	12,196.3
Panama	46	7.2	34.8	28.3	113.0
Papua N.G.	54	60.5	5.6	18.5	160.5
Peru	131	595.9	21.4	22.1	1,248.6
Philippines	450	123.7	22.0	14.7	1,291.7
Romania	53	48.1	34.0	20.8	2,383.4
South Africa	84	24.0	27.4	7.1	85.6

Table 2. Summary statistics by country

Country	Disasters	Average number of casualties	Percentage hit leader's birth region	Percentage funded	Average funding in 1,000s (2017 USD)
Sri Lanka	73	55.3	41.1	23.3	456.9
Sudan	75	118.7	18.7	22.7	1,063.0
Tajikistan	57	9.7	43.9	28.1	543.0
Tanzania	70	64.8	8.6	10.0	64.2
Thailand	99	41.6	12.1	15.2	223.0
Turkey	121	277.8	14.0	12.4	3,733.1
Uganda	61	30.6	4.9	11.5	122.4
Venezuela	39	794.7	10.3	17.9	564.5
Vietnam	178	87.6	11.2	14.6	283.4
All disasters	6,228	360.0	18.9	13.4	1,776.5

Table 2. Summary statistics by country (continued)

Notes: Average funding includes only aid provided by OFDA (rather than all US government agencies). We exclude disasters without any funding from this average.

We further provide an overview of descriptive statistics for all countries in Table 2.²⁰ Overall, there is substantial heterogeneity in the indicators. The number of disasters included in the sample ranges from 37 in El Salvador to 748 in China, while the percentage of funded disasters varies between 4.1% (China) and 52% (Costa Rica).²¹ There are further significant differences in the percentages of disasters hitting the leader's birth region, with values ranging from 4.8% (Mexico) to 47.3% (Haiti). Finally, the included disasters are on average deadliest in Myanmar, with a mean of 2,569.9 casualties.

4. Empirical strategy

We estimate a range of regression models to test for birth region-related favoritism in the provision of disaster relief. We follow Fink and Redaelli (2011) and Raschky and Schwindt (2012) by estimating this relationship as a two-step process. The first stage of the aid allocation model captures whether or not a disaster is funded; our dependent variable Y_{ect} is thus binary and takes the value of one if any aid is granted for disaster *e* that took place

²⁰ We provide descriptive statistics for the control variables in Table A1.

²¹ These figures do not involve funding practices of the United States for disaster types that are not included in this analysis. Especially for African countries, complex emergencies and droughts are salient and receive generous support from the United States.

in year *t* in recipient country *c*. In the second stage, the outcome is the amount of OFDA aid that was disbursed for a certain disaster, in logged 2017 US Dollars. Given the structure of the dataset, we estimate regressions at the disaster level rather than at the regional level. As a starting point, our empirical models are thus of the following form:

$$Y_{ect} = \beta_1 birth_{ect} + \beta_2 X_{ect} + \gamma_c + \theta_e + \delta_t + \varepsilon_{ect}.$$
 (1)

In all specifications, *birth_{ect}* is a binary indicator that is equal to one if any ADM1 area hit by a disaster is the birth region of the leader governing the country at the time of the disaster, and zero otherwise. The vector of control variables X_{ect} captures local area characteristics as discussed in Section 2. Note that while some of the controls are constructed with time-invariant input data, they still vary between country-specific disasters to the extent that disasters hit different areas. Moreover, we control for fixed effects on the country (γ_c), disaster-type (θ_e) and year (δ_t) levels. In a further specification, we add the number of casualties reported by EM-DAT as a proxy for disaster magnitude. This variable should be interpreted with caution, as even though we are limiting the focus to rapid-onset disasters, the number of casualties might arguably be endogenous to disaster aid.²² We cluster standard errors at the country level.

In a more restrictive model, we then replace all control variables by disaster-area fixed effects, thus restricting the analysis to disasters that hit the same ADM1 areas with alternating birth region status over time. By comparing disasters hitting the exact same ADM1 areas, this specification brings the major advantage of allowing us to control for all factors that do not vary at the ADM1-level over time. That being said, the inclusion of restrictive fixed effects can reduce the signal-to-noise ratio, making results more sensitive to individual observations.

²² Given the potential endogeneity of disaster magnitude, we do not use this variable in Sections 6 and 7. Results are however highly similar and are available on request.

As a third identification strategy, we therefore introduce a placebo model with an alternative exposure variable (henceforth "placebo") taking the value of one for disasters that hit regions that were birth regions in the previous or subsequent period but not during the disaster, and zero otherwise.²³ We thereby exploit a discontinuity introduced by leader changes. While new leaders may have been born in a different region than their predecessors, introducing a sudden change in the birth region status of affected regions, changes in unobserved confounding variables are unlikely to simultaneously exhibit discontinuities even if they vary over time. Regions that are the birthplace of a leader who just lost power or will gain power in the very near future should therefore exhibit the same underlying traits as the contemporaneous leader's birth region. The model is hence able to control for both time-invariant and time-variant unobserved heterogeneity.²⁴ A statistically significant coefficient for the placebo would thus indicate that our identification strategy might suffer from omitted variables bias. Differences between our variable of interest and the placebo can be understood as the bias-corrected net effect of disasters hitting the birth region of the current leader.²⁵

5. Main results

We begin by examining whether disasters hitting the birth region of the country's leader have a higher probability of receiving humanitarian aid than disasters hitting non-birth regions. In Table 3, the dependent variable in all columns takes the value of one if the disaster received any aid and zero otherwise.

²³ We define previous and subsequent periods as one year before the start date and one year after the end date of disasters, respectively. For observations with missing days or days and months, we assumed the 15th of each month and the 1st of July as start and end dates, respectively.

²⁴ This placebo model is particularly powerful in the context of rapid-onset natural disasters, given their random timing.

²⁵ This interpretation abstracts from the possibility that leaders who just lost or are about to gain de jure power may already/still hold a certain degree of de facto power. Therefore, the placebo model, strictly speaking, provides a falsification test, as, with sufficient power, a lack of significance for the placebo coefficient rules out the discussed form of omitted variable bias but a significant coefficient does not prove with full certainty that omitted variable bias is present.

Birth region 0.050 ^{***} (0.015) 0.022 (0.014) 0.026 ^{**} (0.013) 0.027 ^{**} (0.013) -0.031 (0.013) Economic/strategic importance -0.001 0.001 ^{**} (0.000) 0.001 ^{**} (0.000) 0.001 ^{**} (0.000) # ports 0.032 ^{**} (0.013) 0.022 0.022 ^{**} (0.014) 0.001 ^{**} (0.000) # nuclear plants -0.006 ^{***} (0.013) -0.003 ^{**} (0.011) 0.001 ^{**} (0.011) Capital city 0.044 ^{***} (0.013) 0.044 ^{***} (0.014) 0.046 ^{***} (0.041) Nightlight p.c. -0.005 ^{**} (0.013) -0.007 ^{***} (0.002) -0.006 ^{***} (0.004) Population 0.005 ^{***} (0.002) -0.007 ^{***} (0.002) -0.006 ^{***} (0.002) Accessibility Ruggedness -0.008 ^{**} (0.004) -0.002 ^{**} (0.004) -0.002 ^{**} (0.004) Magnitude # deaths (in 1,000s) X X X X Disaster type Year Disaster area X X X X Observations 6.228 6.228 6.228 6.228 6.228		(1) Any funding	(2) Any funding	(3) Any funding	(4) Any funding	(5) Any funding
Economic/strategic importance % barren land 0.000 0.001' 0.001' 0.001' 0.001' 0.001' 0.001' 0.001' 0.001' 0.001' 0.001' 0.001' 0.001' 0.001' 0.001' 0.001' 0.001' 0.001' 0.001' 0.001' 0.001' 0.001' 0.001' 0.001' 0.001' 0.001' 0.001' 0.001' 0.001' 0.001' 0.001' 0.001' 0.001' 0.001' 0.001' 0.001' 0.001' 0.001' 0.003'' 0.0003'' 0.0003'' 0.0003'' 0.003'' 0.003''' 0.003''' 0.003''' 0.003''' 0.003''' 0.003''' 0.003''' 0.001'' 0.014''' 0.046'''' 0.046'''' 0.004'''' 0.004'''' 0.004'''' 0.001''''' 0.001''''' 0.002'' 0.002'' 0.002'' 0.002'' 0.002'' 0.002'' 0.002'' 0.002'' 0.002''' 0.002''' 0.002''' 0.002'''' 0.002'''' 0.002'''' 0.002'''' 0.002'''' 0.002'''' 0.002'''' <t< td=""><td>Birth region</td><td>0.050*** (0.015)</td><td>0.022 (0.014)</td><td>0.026[*] (0.013)</td><td>0.027** (0.013)</td><td>-0.031 (0.030)</td></t<>	Birth region	0.050*** (0.015)	0.022 (0.014)	0.026 [*] (0.013)	0.027** (0.013)	-0.031 (0.030)
% barren land 0.000 0.001* 0.001* % barren land (0.000) (0.000) (0.000) # ports 0.032" 0.022* 0.022* (0.013) (0.011) (0.011) (0.011) # nuclear plants -0.006** -0.003* -0.003* (0.02) (0.002) (0.001) (0.014) Nightlight p.c. 0.044** 0.048*** 0.046*** Nightlight p.c. -0.070* -0.029 -0.028 (0.041) (0.047) (0.047) (0.047) Population -0.005** -0.007** -0.006*** Pop. density -0.005** -0.011*** 0.011*** O.002) (0.002) (0.002) (0.002) City population 0.005** -0.011*** 0.011*** Ruggedness -0.008* -0.003 -0.002 Magnitude # deaths (in 1,000s) 0.004*** (0.004) Fixed effects Country X X X Disaster area X<	Economic/strategic importance					
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	% barren land		0.000 (0.000)	0.001 [*] (0.000)	0.001 [*] (0.000)	
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	# ports		0.032** (0.013)	0.022 [*] (0.011)	0.022 [*] (0.011)	
Capital city 0.044*** 0.048*** 0.046*** Nightlight p.c. -0.070* -0.029 -0.028 Population (0.041) (0.047) (0.047) Pop. density -0.005** -0.007*** -0.006*** City population 0.005*** 0.011*** 0.011*** Ruggedness -0.008** -0.002 (0.002) Accessibility -0.008** -0.003 -0.002 Magnitude -0.004** (0.004) (0.004) # deaths (in 1,000s) 0.004*** (0.001) Fixed effects 0.004*** (0.001) Country X X X Year X X X Disaster type X X X Year X X X X Observations 6.228 6.228 6.228 6.228 6.228	# nuclear plants		-0.006*** (0.002)	-0.003 [*] (0.002)	-0.003 [*] (0.001)	
Nightlight p.c. -0.070° -0.029 -0.028 (0.041) (0.047) (0.047) Population -0.005 ^{**} -0.007 ^{***} -0.006 ^{***} Pop. density -0.002) (0.002) (0.002) City population 0.005 ^{***} 0.011 ^{***} 0.011 ^{***} City population 0.002) (0.002) (0.002) Accessibility -0.008* -0.003 -0.002 Ruggedness -0.008* -0.003 -0.002 Magnitude (0.004) (0.004) (0.004) # deaths (in 1,000s) 0.004 ^{***} (0.001) Fixed effects 0.004 ^{***} (0.001) Fixed effects X X X Country X X X Year X X X Disaster type X X X Year X X X Disaster area X X X Qbservations 6.228 6.228 6.228 6.228	Capital city		0.044* ^{**} (0.013)	0.048*** (0.014)	0.046*** (0.014)	
Population -0.005** -0.007*** -0.006*** City population 0.002 (0.002) (0.002) Accessibility 0.005*** 0.011*** 0.011*** Ruggedness -0.008* -0.003 -0.002 Magnitude (0.004) (0.004) (0.004) # deaths (in 1,000s) 0.004*** (0.001) Fixed effects 0.004*** (0.001) Country X X X Disaster type X X X Vear X X X Observations 6.228 6.228 6.228 6.228	Nightlight p.c.		-0.070* (0.041)	-0.029 (0.047)	-0.028 (0.047)	
Pop. density -0.005 ^{**} -0.007 ^{***} -0.006 ^{***} City population 0.005 ^{***} 0.011 ^{***} 0.011 ^{***} Ruggedness -0.008 ^{**} -0.003 -0.002 Accessibility -0.008 ^{**} -0.003 -0.002 Magnitude (0.004) (0.004) (0.004) # deaths (in 1,000s) 0.004 ^{***} (0.001) Fixed effects 0.004 ^{***} (0.001) Country X X X Year X X X Disaster area X X X Observations 6.228 6.228 6.228 6.228	Population					
City population (0.002) 0.005 ^{***} (0.002) (0.002) 0.011 ^{***} (0.002) (0.002) (0.002) Accessibility Ruggedness -0.008 [*] (0.004) -0.003 (0.004) -0.002 (0.004) Magnitude # deaths (in 1,000s) -0.003 (0.004) -0.004 ^{***} (0.001) Fixed effects Country Disaster type Year X X X Disaster type Year X X X X Observations 6.228 6.228 6.228 6.228	Pop. density		-0.005**	-0.007***	-0.006***	
City population 0.005 ^{***} 0.011 ^{***} 0.011 ^{***} Accessibility (0.002) (0.002) (0.002) Accessibility -0.008 [*] -0.003 -0.002 Magnitude (0.004) (0.004) (0.004) # deaths (in 1,000s) 0.004 ^{***} (0.001) Fixed effects 0.004 ^{***} (0.001) Country X X X Disaster type X X X Year X X X Disaster area X X X Observations 6.228 6.228 6.228 6.228			(0.002)	(0.002)	(0.002)	
Accessibility (0.002) (0.002) (0.002) Ruggedness -0.008* -0.003 -0.002 Magnitude (0.004) (0.004) (0.004) # deaths (in 1,000s) 0.004*** (0.001) Fixed effects 0.004*** (0.001) Fixed effects X X X Country X X X Pisaster type X X X Year X X X Disaster area X X X Observations 6.228 6.228 6.228 6.228	City population		0.005***	0.011***	0.011***	
Accessibility -0.008* -0.003 -0.002 Ruggedness (0.004) (0.004) (0.004) Magnitude # deaths (in 1,000s) 0.004*** (0.001) Fixed effects 0.0011 (0.001) 0.004*** Country X X X X Disaster type X X X X Year X X X X Disaster area 6.228 6.228 6.228 6.228			(0.002)	(0.002)	(0.002)	
Ruggedness -0.008° (0.004) -0.002 (0.004) Magnitude (0.004) (0.004) # deaths (in 1,000s) 0.004 ^{***} (0.001) Fixed effects 0.001) Country X X X Disaster type X X X Year X X X Disaster area X X X Observations 6.228 6.228 6.228 6.228	Accessibility					
Magnitude0.004***# deaths (in 1,000s)0.004***Fixed effects0.001)CountryXXDisaster typeXXYearXXDisaster areaXObservations6.2286.2286.2286.2286.228	Ruggedness		-0.008 [*] (0.004)	-0.003 (0.004)	-0.002 (0.004)	
# deaths (in 1,000s) 0.004*** Fixed effects (0.001) For the country X X X Disaster type X X X Year X X X Disaster area X X X Observations 6.228 6.228 6.228 6.228	Magnitude		· · ·	· · · ·	· · · ·	
Fixed effectsCountryXXXXDisaster typeXXXYearXXXDisaster areaXXXObservations6.2286.2286.2286.228	# deaths (in 1,000s)				0.004*** (0.001)	
CountryXXXXDisaster typeXXXYearXXXDisaster areaXXObservations6.2286.2286.228	Fixed effects				· · · ·	
Disaster typeXXXYearXXXDisaster areaXXObservations6.2286.2286.228	Country	Х	Х	Х	Х	
YearXXDisaster areaXXObservations6.2286.2286.228	Disaster type		Х	Х	Х	
Disaster area X Observations 6.228 6.228 6.228 6.228	Year			Х	Х	
Observations 6.228 6.228 6.228 6.228 6.228	Disaster area					Х
D^2_{2} 0.067 0.114 0.192 0.199 0.619	Observations	6,228	6,228	6,228	6,228	6,228

Table 3. Birth region effect on the probability to receive humanitarian aid

Notes: ***, **, ** denote statistical significance at the 1, 5 and 10% level respectively, with clustered standard errors in parentheses. The sample consists of 50 countries over the 1964-2017 period. The dependent variable is a binary indicator taking the value of one if the disaster was granted any funding and zero otherwise. The variable birth region takes the value of one for disasters that hit the birth region of the country leader and zero otherwise.

In column 1, which adjusts for country-level fixed effects only, we find that a disaster hitting the leader's birth region significantly increases the probability to receive aid by five percentage points. However, this association turns insignificant after controlling for local area characteristics related to economic/strategic importance, population, and accessibility as well as for disaster-type fixed effects (column 2).²⁶ The birth region coefficient regains statistical significance once year fixed effects (column 3) and disaster magnitude (column 4) are accounted for, and reaches a magnitude of 2.6 and 2.7 percentage points, respectively. Finally, in column 5, we apply our most conservative model using disaster-area fixed effects.²⁷ In this specification, the coefficient is not statistically significant and turns negative. In summary, our estimates show that there is no robust evidence for systematic favoritism in the probability to receive any aid.

We next turn to the birth region effect on aid amounts in logged 2017 US-Dollars. This analysis draws upon the reduced sample of disasters with positive aid flows, which covers all 50 countries and 13.4% of all disasters in our dataset. Following the same structure as Table 3, Table 4 shows that disasters hitting leaders' birth regions receive substantially higher aid amounts than other disasters. Estimates are robust to controlling for local area characteristics, disaster-type fixed effects, year fixed effects, and disaster magnitude (columns 2 to 4). Across these specifications, the statistically significant increase in funding is of approximately 45%-85%, on average.²⁸ For the average disaster's funding in our sample, this corresponds to an increase by US\$849,984 to US\$1,506,135.

²⁷ Given that this model reduces the variation of interest to a much narrower set of disasters, we do not include year fixed effects or other controls in order to maintain a sufficient number of degrees of freedom.

²⁶ The size of the population residing in cities, the number of ports and the fact that a disaster hit a nation's capital city are positively associated with the probability to receive disaster assistance across specifications. In contrast, countries are less likely to receive humanitarian aid the higher the population density and the more nuclear plants are located in disaster-affected areas. While the control variables condition on each other and may not represent causal estimates, these results suggest that nuclear plants may introduce a different geopolitical dynamic and it may not always be in the interest of the United States to extend aid if these are located in disaster areas.

²⁸ Given that the outcome is logged US Dollars, the marginal effect for the binary birth region indicator expressed as percentage change in funding can be calculated as $\frac{Y(1)-Y(0)}{Y(0)} = e^{\beta_1} - 1$, where Y(1) and Y(0) represent the outcome with the indicator switched on and off, respectively, and β_1 being the estimated coefficient. A potential concern relates to the second stage of the aid allocation decision being subject to selection bias given that the aid amounts regression is estimated on a subsample of regions hit by disasters. As we lack a clear candidate for an excludable instrument for the first stage, the performance of the conventionally used Heckman Selection Model is limited. When we estimate a Heckman Selection Model without exclusion restriction most results are however unchanged (results available on request).

	(1)	(2)	(3)	(4)	(5)
	Log	Log	Log	Log	Log
	funding	funding	funding	funding	funding
Birth region	0.614***	0.501***	0.368**	0.391**	0.941 ^[*]
	(0.156)	(0.155)	(0.149)	(0.147)	(1.104)
					[0.057]
importance					
% barren land		0.010**	0.005	0.009***	
		(0.004)	(0.004)	(0.003)	
# ports		-0.353**	-0.162	-0.146	
•		(0.175)	(0.151)	(0.155)	
# nuclear plants		-0.190***	-0.195***	-0.164***	
•		(0.057)	(0.054)	(0.060)	
Capital city		0.169 [´]	0.017	-0.003	
		(0.155)	(0.152)	(0.149)	
Nightlight p.c.		0.143	1.307	`1.162 [´]	
0 0 1		(3.924)	(3.716)	(3.607)	
Population		· · ·	· · · ·	· · ·	
Pop. density		0.006	0.006	0.006	
		(0.032)	(0.044)	(0.044)	
City population		Ò.314* ^{***}	Ò.267* ^{***}	Ò.235* ^{***}	
		(0.049)	(0.046)	(0.049)	
Accessibility		· · ·	· · · ·	· · · ·	
Ruggedness		-0.083	-0.116**	-0.101*	
55		(0.055)	(0.053)	(0.053)	
Magnitude		· · ·	· · · ·	· · ·	
# deaths (in 1,000s)				0.024***	
				(0.004)	
Fixed effects				· · ·	
Country	Х	Х	Х	Х	
Disaster type		Х	Х	Х	
Year			Х	Х	
Disaster area					Х
Observations	836	836	836	836	836
R^2	0.151	0.227	0.343	0.381	0.881

Table 4. Birth region effect on amount of received aid

Notes: ***, ** denote statistical significance at the 1, 5 and 10% level respectively. Stars in brackets denote statistical significance based on clustered wild bootstrapping. Clustered standard errors in parentheses. Column 5 displays p-value based on clustered wild bootstrapping in brackets. The sample consists of 50 countries over the time period 1964-2017. The dependent variable is the disaster-specific log funding in 2017 US dollars, conditioning on having received any funding. The variable birth region takes the value of one for disasters that hit the birth region of the country leader and zero otherwise.

While these specifications already acknowledge important confounders, substantially reducing the scope of alternative interpretations for the birth region coefficient, they control for the heterogeneity of disaster-affected areas imperfectly.²⁹ In Column 5, we therefore introduce disaster-area fixed effects, which fully absorb all time-invariant heterogeneity

²⁹ As before, we find – consistently across specifications – that the number of nuclear plants is negatively associated with funding, whereas the size of the population residing in cities increases aid.

between disaster-affected areas. Note that in this restrictive model, the number of countries contributing to the point estimate is reduced to a subset of 14 countries that were affected by disasters hitting the exact same ADM1 areas multiple times with alternating birth region status. As can be seen, the coefficient is not statistically significant due to a substantial loss in precision. However, clustered standard errors are not appropriate in regressions with such a small number of clusters. For this reason, we also report a p-value based on clustered wild bootstrapping, which are more suitable for this setting (see Roodman et al., 2019). As can be seen, the birth region effect is significant at the ten percent level and increases to 156%. Interestingly, countries exhibiting this type of variation score worse in terms of governance indicators that are arguably associated with higher degrees of favoritism, likely driving the observed increase in effect size.³⁰ Taken together, Table 4 lends strong support for the existence of substantial favoritism.

Tables 3 and 4 suggest the absence of birth region favoritism at the extensive margin but a strong presence of it at the intensive margin. This combined result may be rooted in the decision-making process within OFDA. As discussed in more detail in Section 6, US ambassadors have the authority to grant small amounts of aid on behalf of OFDA as an initial response to disasters. This type of decision is made quickly after disaster onset, when the full geographic spread of a disaster may not yet be fully understood. Hence, it may not leave much room for complex strategic considerations that would be reflected in regional favoritism. In cases where the ambassador's rapid-response assistance is deemed insufficient, more detailed impact assessment and coordination take place to determine further steps, creating substantial room for the formation of favoritism.

We next perform a series of placebo tests in Table 5 that exploit discontinuities of birth region status over time and the random timing of rapid-onset disasters to further ease endogeneity concerns. We reproduce columns 1 and 2 from Tables 3 and 4 with the addition of the placebo indicator.³¹ Results confirm the absence of systematic favoritism

³⁰ The indicators we investigated are extent of clientelism in public spending, the quality of the local bureaucracy, and government accountability. A more detailed discussion of these indicators is provided in Section 6.

³¹ Since this analysis exploits variation in the timing of disasters, we do not include year fixed effects to ensure sufficient power.

at the extensive margin. The placebo indicator is not only positive and significant but also exceeds the birth region indicator in size in both specifications. In stark contrast to this, the placebo estimates in the aid amounts regressions are negative and insignificant, while the birth region coefficients themselves remain positive and highly significant. This is further evidence that birth region effects are attributed to the political power of affected areas, rather than being driven by unobserved inherent characteristics of birth regions.

	(1) Any funding	(2) Any funding	(3) Log funding	(4) Log funding
	Any funding	Any funding	Log runuing	Log runuing
Birth region	0.053***	0.024*	0.608***	0.480***
Placebo	0.077***	0.063**	-0.073	-0.157
	(0.020)	(0.023)	(0.230)	(0.270)
Controls		Х		Х
Fixed effects				
Country	Х	Х	Х	Х
Disaster type		Х		Х
Bias-corrected p-value	n/a	n/a	0.0001	0.0002
Direct test (p-value)	n/a	n/a	0.0159	0.0373
Observations	6,161	6,161	836	836
R^2	0.068	0.115	0.151	0.227

Table 5. Placebo models

Notes: ***, **, * denote statistical significance at the 1, 5 and 10% level respectively, with clustered standard errors in parentheses. The table replicates columns 1 and 2 from Tables 3 and 4 with the addition of the placebo indicator. The underlying test statistics for the bias-corrected p-values are calculated by subtracting the placebo coefficient from the birth region coefficient and dividing by the standard error of the birth region coefficient. The direct tests consider the Null hypothesis that the birth region and placebo coefficients are equal.

To further substantiate this point, we conduct two formal tests. First, we divide the difference between the birth region and placebo coefficients by the standard error of the birth region estimate in order to test whether the bias-adjusted birth region coefficient remains statistically significant. As shown by the bias-corrected p-values at the bottom of Table 5, this is clearly the case (with p-values < 1%). A potential problem with this approach is that the placebo estimates may suffer from a lack of power and only happen to be close to zero. We therefore conduct a second more conservative test, in which we directly test whether the birth region and placebo estimates are significantly different from each other. The results are reassuring. This more conservative test also yields statistically

significant results (with p-values <5%). Therefore, the placebo analysis suggests that our results cannot be explained by characteristics inherent to potential birth regions.

Taken together, the previous analyses thus demonstrate that there is strong evidence in line with regional favoritism at the intensive margin while we do not find systematic evidence for the decision of whether or not to grant aid in the first place. Before discussing further extensions to the main regression models, we proceed with analyzing potential factors facilitating the observed relationship.

6. Do US-interests facilitate the home bias?

In this section we test whether the birth region-effect we have uncovered is actively supported by the United States or is exclusively the result of recipient country politics, in tandem with the absence of sufficient safeguards to prevent such favoritism. To discern how regional favoritism can enter the decision process on humanitarian aid amounts, it is useful to first review how the OFDA response to disasters unfolds. As mentioned earlier, the disaster response usually begins with a US diplomat, either a US Ambassador, Chief of Mission or US Assistant Secretary of State, responding to an assistance request from the recipient country's government.³² This occurs when the disaster magnitude is declared to exceed the response capacity of affected countries. Afterwards, usually within 24 hours, the US diplomat can allocate a limited amount of up to US\$ 50,000 (OFDA, 2010; Margesson, 2013, Kevlihan et al., 2014).³³

OFDA officials coordinate with the government of the affected countries to determine whether and how much additional aid should be granted. Importantly, while the level of coordination can vary, the governments of affected countries are usually involved in the needs assessments by OFDA, which serve as a basis for determining additional aid amounts. The involvement of recipient countries can be either passive, by providing information on the disaster damage, or active, by participating in joint assessments with

³² Alternatively, recipient countries can also accept an offer from US representatives.

³³ Before 2003, the limited amount was US\$ 25,000 (Kevlihan et al., 2014).

OFDA teams. The OFDA teams deployed to the affected areas to carry out these assessments vary in size. For instance, a five-person team was sent to affected areas by typhoon Megi in the Philippines in 2010. The team's objective was to identify humanitarian needs together with officials of the Government of the Philippines, among others. The assessment was later used as a basis to allocate additional funds amounting to US\$ 1.1 million (OFDA, 2011). For the case of large-scale disasters, bigger elite missions of humanitarian experts and technical advisors are deployed to the affected areas.³⁴ For example, the team for Hurricane Matthew affecting Haiti, Jamaica and the Bahamas in 2016 surpassed 70 members at its peak. Moreover, OFDA can grant additional aid without the deployment of teams to the affected areas, in which case information provided by local governments further gains in importance (OFDA, 2008-2017; Margesson, 2013, Kevlihan et al., 2014).

The involvement of the government in damage assessments, as well as the difficulties in verifying all governmental information, could enable country leaders to favor disasters hitting their birth regions in multiple ways. First, assessments might become more accurate for such disasters, improving their quality and increasing the likelihood of higher aid amounts, while no such care might be taken when non-birth regions are hit. Second, both the number of casualties and physical damage could be intentionally magnified for disasters hitting birth regions. Third, if non-birth regions are more likely to be politically misaligned with the country leader, disasters affecting those regions could be intentionally underplayed, or governments could hinder the entry of humanitarian aid to the country.³⁵

As our setting puts the relationship between the population in disaster-affected areas and the country leader at the center of the analysis, it is natural to argue that leader's interests are at work. Following this logic, leaders might favor their birth regions for two reasons. They might simply derive utility from supporting a community they identify with, or attempt

³⁴ In the US fiscal year of 2017, a record of 6 out of 53 disasters with OFDA funding required such missions, referred to as disaster assistance response teams (DARTs, see OFDA, 2017).

³⁵ For instance, the government of Burma was criticized for blocking and delaying the entry of humanitarian aid for the damages caused by cyclone Nargis in 2008 (Martin and Margesson, 2008; OFDA, 2008).

to ensure electoral and political support from their stronghold (Hodler and Raschky, 2014; Do et al., 2017; Anaxagorou et al., 2019; Dreher et al., 2019).

Nevertheless, the political and economic interests of the United States might also be at play and facilitate additional funding for disasters affecting the birth region of leaders. In this section, we focus on two dimensions that we expect to increase the US government's stakes in a recipient country. The first is geopolitical alignment, which is often proxied with a country's voting behavior in the United Nations General Assembly (e.g., Fuchs and Klann, 2013). Using this proxy, Faye and Niehaus (2012) show that the political interests of donor governments distort the allocation of aid towards their allies in election years, with the aim to help such allies stay in power.³⁶ Requests from leaders aligned with the United States might then be more likely to be approved. In order to test whether the United States selectively allocates its humanitarian aid towards its allies, we interact our birth region-indicator with a (lagged) measure for the share of votes in which the disaster-affected countries were in agreement with the United States in the UNGA.³⁷

The second dimension we focus on is a recipient country's commercial relations to the United States. A substantial literature explores how trade can be used to punish or reward countries (Berger et al., 2013; Fuchs and Klann, 2013), potentially with the aim to harm or support incumbent governments. We capture economic ties by using dyadic trade data from the Correlates of War project (Barbieri et al., 2009; Barbieri and Keshk, 2016) and calculate the sum of (lagged) US imports from and exports to disaster-affected countries both as shares of US GDP and in absolute terms. Again, to the extent that the US government grants humanitarian aid with the aim to support the incumbent of the recipient country in mind, more aid should be granted to birth regions if the recipient's trade ties with the US are stronger.

³⁶ Also see Rommel and Schaudt (2017).

³⁷ We take UN voting data from Voeten (2013), relying on the fraction of votes for which the United States was in agreement with the respective recipient country (counting abstentions as half-agreement).

	(1)	(2)	(3)
	Log funding	Log funding	Log funding
Birth region	0.220	0.419**	0.424**
-	(0.364)	(0.171)	(0.168)
Birth region X UNGA	0.516	. ,	. ,
overlap	(0.988)		
Birth region X trade/GDP	· · · ·	-1.059	
C C		(0.690)	
Birth region X trade			-0.009
(absolute)			(0.005)
Controls	Х	Х	Х
Fixed effects			
Country	Х	Х	Х
Disaster type	Х	Х	Х
Year	Х	Х	Х
Observations	806	823	823
R^2	0.339	0.335	0.335

Table 6. Birth region effect and US interests

Notes: ***, **, * denote statistical significance at the 1, 5 and 10% level respectively, with clustered standard errors in parentheses. The dependent variable is the disaster-specific log funding in 2017 US dollars, conditioning on having received any funding. The variable birth region takes the value of one for disasters that hit the birth region of the country leader and zero otherwise. Absolute trade values are scaled to 2017 Billion US dollars. The main effects of the interacted variables are included in the regressions.

We show the results in Table 6. As can be seen, neither political nor commercial motives seem to be at play. Column 1 shows no evidence to support the idea of political interests exacerbating or ameliorating regional favoritism. According to columns 2 and 3, trade with the United States does not significantly affect the degree of favoritism in humanitarian aid, suggesting that US economic interests do not influence the observed effect. Overall, the results thus show no evidence in support of the hypothesis that the United States uses its humanitarian aid to facility requests from closer allies over those from more distant countries, in terms of geo-political and commercial relations.

In order to further test the potential role of the United States in the birth region-effect, we focus on a number of variables that we expect to affect how the US evaluates appeals. According to Eisensee and Strömberg (2007), the US government is particularly keen to support countries hit by disasters when they receive more media attention in the US, arguably because it pleases their own support base. We thus focus on US media attention to natural disasters, exploiting information stored in the Vanderbilt Television News Archive (n.d.), and create a binary indicator capturing whether a disaster was featured at

least once during the evening news of the ABC news network.³⁸ When media attention is larger the US government will find it harder to support its allies. At the same time, media attention might increase US scrutiny for favoritism-based requests from the recipient, which would also reduce the home bias-effect.

We similarly expect the US government to be more skeptical of politically charged requests for help in countries with pronounced clientelism, low bureaucratic quality, and weak governance, as these are arguably those environments where the abuse of funds is most likely. We also test whether the US is more attentive at (recipient country) election time, as it is in such years that recipient country leaders might be most keen to misrepresent need and channel additional funds to their birth region (Anaxagorou, et al. 2019; Dreher et al., 2019).

For this exercise, we use a continuous index from the Variety of Democracy (V-Dem) project, in which higher scores stand for a lower degree of clientelism in government spending (i.e., a larger fraction of social and infrastructure expenditures are public goods rather than favoring particularistic interests). This indicator is based on country expert assessments and covers all countries and years in our sample (Coppedge et al., 2018). Second, we make use of two measures for quality of bureaucracy and democratic accountability obtained from the International Country Risk Guide/ICRG (ICRG, 2013). The former captures the institutional strength and autonomy of local bureaucracies on a scale from 0 to 4, with higher values indicating that the bureaucracy is able to govern without drastic policy changes or service interruptions (for instance, thanks to established staff recruiting systems and administrative procedures). The latter measure ranks countries on a scale from 0 to 6 taking into consideration the degree of political competition as well as checks and balances. Again, higher values indicate higher quality. Unlike the V-Dem data, the ICRG measures are not based on expert judgement but follow

³⁸ Vanderbilt Television News Archive only began recording on August 5th, 1968, leading to a small loss in the number of observations. We focus on ABC, as other major news networks such as CNN or Fox News started operating at a later date and would hence be unavailable for a large number of disasters. Matching of news reports and disasters is done on the basis of disaster dates as well as news content summaries reported in the Vanderbilt Television News Archive.

predetermined coding rules (see ICRG, 2013 for details).³⁹ Finally, we make use of election data from Cruz et al. (2018) to estimate our models interacted with election indicators. We construct election indicators that take the value of one if an executive, legislative or any of the two type of elections took place within five months after the disaster month.⁴⁰

$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$								
Log fundingLog fundingLog fundingLog fundingLog fundingLog fundingLog fundingLog fundingBirth region 0.357^{**} 0.412^{***} 1.402^{***} 1.204^{**} 0.350^{**} 0.390^{**} 0.365^{**} Birth region X Media -0.121 (0.159) (0.146) (0.290) (0.532) (0.149) (0.147) (0.148) Birth region X Bureaucracy -0.261^{**} (0.144) -0.619^{***} (0.161) -0.215^{**} (0.118) -0.2215^{**} (0.118)Birth region X Bureaucracy -0.619^{***} (0.161) -0.2215^{**} (0.118) -0.288 (0.451) -0.288 (0.451)Birth region X Leg. elections -0.288 (0.451) -0.288 (0.451) -0.288 (0.451)Birth region X Leg. elections X X X X X X X X X X X XFixed effects Country Disaster type X X X X X X X X X X X X X X		(1)	(2)	(3)	(4)	(5)	(6)	(7)
tundingfundingfundingfundingfundingfundingfundingfundingBirth region0.357"0.412"''1.402"''1.204"'0.350"'0.390"'0.365"'Birth region X-0.121(0.146)(0.290)(0.532)(0.149)(0.147)(0.148)Birth region X-0.121-0.261'-0.261'-0.215'-0.215'-0.215'Particularistic(0.144)-0.619"''(0.118)0.668-0.288Birth region X-0.215'-0.215'-0.288(0.451)Birth region X-0.215'-0.288(0.451)Birth region X-0.215'0.3668-0.288Leg. elections-0.218(0.451)0.146Birth region X-0.215'-0.288(0.451)ControlsXXXXXFixed effects		Ĺóa	Ĺóa	Ĺóa	Ĺóg	Ĺóa	Ĺóa	Ĺóg
Birth region 0.357" 0.412"" 1.402"" 1.204" 0.350" 0.390" 0.365" Birth region X -0.121 (0.159) (0.146) (0.290) (0.532) (0.149) (0.147) (0.148) Birth region X -0.121 (0.256) (0.144) (0.144) (0.147) (0.148) Birth region X -0.261" -9articularistic (0.144) (0.161) (0.161) Birth region X -0.215" -0.215" -0.215" -0.288 -0.288 Birth region X -0.215" (0.161) 0.668 (0.451) -0.288 Birth region X -0.288 (0.451) 0.146 (0.475) Birth region X -0.288 (0.451) 0.146 (0.475) Controls X X X X X X Birth region X -0.288 -0.288 (0.451) 0.146 Elections -0.245 -0.288 (0.475) 0.146 Controls X X X		fundina	fundina	fundina	fundina	fundina	fundina	fundina
Birth region 0.357" (0.159) 0.412"' (0.159) 1.402"'' (0.290) 1.204"' (0.532) 0.350"'' (0.149) 0.390"'' (0.147) 0.365"'' (0.148) Birth region X -0.121 (0.256) -0.261" (0.290) (0.532) (0.149) (0.147) (0.148) Birth region X -0.261" -0.619"" -0.215" -0.215" -0.215" -0.215" -0.215" -0.288 -0.288 -0.215" -0.215" -0.288 (0.451) -0.288 -0.245" -0.288 -0.245" -0.288 -0.288 -0.288 -0.245" -0.288 -0.245" -0.288 -0.288 -0.245" -0.245" -0.288 -0.245" -0.288 -0.245" -0.288 -0.288 -0.288 -0.245" -0.288 -0.288 -0.288 -0.245" -0.288 -0.245" -0.245" -0.245" -0.245" -0.245" -0.245" -0.245" -0.245" -0.245" -0.245" -0.245" -0.245" -0.245" -0.245" -0.245" -0.245" -0.245" -0.245" -0.245" -0		. a. rainig		lanang	. an ran rg	lanang	. an raining	<u> </u>
Difference (0.159) (0.146) (0.290) (0.532) (0.149) (0.147) (0.148) Birth region X -0.121 (0.146) (0.290) (0.532) (0.149) (0.147) (0.148) Birth region X -0.261* -0.619*** -0.619*** -0.215* -0.215* -0.215* -0.215* -0.215* -0.215* -0.215* -0.215* -0.215* -0.215* -0.215* -0.215* -0.215* -0.215* -0.215* -0.215* -0.215* -0.215* -0.215* -0.215* -0.215* -0.215* -0.2215* -0.215* -0.215* -0.215* -0.215* -0.215* -0.2215* -0.215* -0.215* -0.2215* -0.215* -0.215* -0.2215* -0.2215* -0.2215* -0.2288 -0.2288 -0.2288 -0.2288 -0.2288 -0.2288 -0.2288 -0.2288 -0.2288 -0.2288 -0.2451) -0.2451) -0.2451) -0.2451) -0.2451) -0.2451) -0.2451) -0.2455 -0.2455 -0.2455 -0.2455 -0.2455 -0.2455 -0.2455 -0.2455 -0.2455 -0.2455 -0.2455	Birth region	0 357**	0 412***	1 402***	1 204**	0 350**	0 300**	0 365**
Birth region X Media-0.121 (0.256)(0.140)(0.140)(0.141)(0.140)Birth region X Particularistic-0.261* (0.144)-0.619*** (0.161)Birth region X Bureaucracy-0.619*** (0.161)-0.215* (0.118)Birth region X Ex. elections-0.215* (0.118)Birth region X Leg. elections-0.215* (0.146)Birth region X Elections-0.215* (0.118)ControlsX XX XX ElectionsX XX ElectionsX XX ElectionsX 	Dirti region	(0 159)	(0.146)	(0.290)	(0.532)	(0 149)	(0.147)	(0.148)
Diminegion X-0.121Media (0.256) Birth region X-0.261*Particularistic (0.144) Birth region X-0.619***Bureaucracy (0.161) Birth region X-0.215*Accountability (0.161) Birth region X-0.215*Accountability (0.118) Birth region X-0.288Leg. elections (0.451) Birth region X0.146Elections (0.475) ControlsXXXXXXXXXXXValue effectsCountryXXXDisaster typeXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXX<	Birth region X	-0 121	(0.140)	(0.200)	(0.002)	(0.140)	(0.147)	(0.140)
Interface (0.250) Birth region X -0.261^{*} Particularistic (0.144) Birth region X -0.619^{***} Bureaucracy (0.161) Birth region X -0.215^{*} Accountability (0.161) Birth region X (0.668) Ex. elections (0.610) Birth region X -0.288 Leg. elections (0.451) Birth region X 0.146 Leg. elections (0.475) ControlsXXXXXXXXXXXFixed effects $Country$ CountryXXXXXDisaster typeXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXX <td>Modia</td> <td>(0.256)</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>	Modia	(0.256)						
Dirth region X-0.201Particularistic(0.144)Birth region X-0.619***Bureaucracy(0.161)Birth region X-0.215*Accountability(0.118)Birth region X0.668Ex. elections(0.610)Birth region X0.668Leg. elections(0.451)Birth region X0.146Leg. elections(0.451)ControlsXXXXXXXXXXXXXXXFixed effectsCountryCountryXXXXXXXDisaster typeXXXXXXXXXXXXX	Dirth rogion V	(0.250)	0.261*					
Particularistic(0.144)Birth region X-0.619***Bureaucracy(0.161)Birth region X-0.215*Accountability(0.118)Birth region X0.668Ex. elections(0.610)Birth region X-0.288Leg. elections(0.451)Birth region X0.146Leg. elections(0.475)ControlsXXXXXFixed effectsCountryCountryXXXXXDisaster typeXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXX <td< td=""><td></td><td></td><td>-0.201</td><td></td><td></td><td></td><td></td><td></td></td<>			-0.201					
Birth region X-0.619Bureaucracy(0.161)Birth region X-0.215*Accountability(0.118)Birth region X0.668Ex. elections(0.610)Birth region X-0.288Leg. elections(0.451)Birth region X0.146Leg. elections0.146ControlsXXXXXXXXXXState effectsCountryCountryXXXDisaster typeXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXX	Particularistic		(0.144)	0.04.0***				
Bureaucracy (0.161) Birth region X -0.215* Accountability (0.118) Birth region X 0.668 Ex. elections (0.610) Birth region X -0.288 Leg. elections (0.451) Birth region X 0.146 (0.451) Controls X X X X X X X X X Fixed effects Country X X X X X X X X X Disaster type X X X X X X X X X	Birth region X			-0.619				
Birth region X Accountability-0.215 (0.118)Birth region X Ex. elections0.668 (0.610)Birth region X Leg. elections-0.288 (0.451)Birth region X Elections0.146 (0.475)ControlsXXXXXXXFixed effects CountryXXXXXXXXDisaster typeXXXX	Bureaucracy			(0.161)	o o / =*			
Accountability(0.118)Birth region X0.668Ex. elections(0.610)Birth region X-0.288Leg. elections(0.451)Birth region X0.146Elections(0.475)ControlsXXXXXXFixed effectsCountryCountryXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXX <td>Birth region X</td> <td></td> <td></td> <td></td> <td>-0.215</td> <td></td> <td></td> <td></td>	Birth region X				-0.215			
Birth region X0.668Ex. elections(0.610)Birth region X-0.288Leg. elections(0.451)Birth region X0.146Elections(0.475)ControlsXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXX <t< td=""><td>Accountability</td><td></td><td></td><td></td><td>(0.118)</td><td></td><td></td><td></td></t<>	Accountability				(0.118)			
Ex. elections(0.610)Birth region X Leg. elections-0.288 (0.451)Birth region X Elections0.146 (0.475)ControlsXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXDisaster typeXXXXXXXXXXXXXXXXXXX	Birth region X					0.668		
Birth region X Leg. elections-0.288 (0.451)Birth region X Elections0.146 (0.475)ControlsXXXFixed effects CountryXXXXXXFixed effects Disaster typeXXXXXX	Ex. elections					(0.610)		
Leg. elections(0.451)Birth region X Elections0.146 (0.475)ControlsXXXXFixed effects CountryXXXXXXDisaster typeXXXXXX	Birth region X						-0.288	
Birth region X Elections0.146 (0.475)ControlsXXXXFixed effects CountryXXXXXXDisaster typeXXXXXX	Leg. elections						(0.451)	
Elections (0.475) Controls X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X	Birth region X							0.146
ControlsXXXXXXXFixed effectsCountryXXXXXXDisaster typeXXXXXX	Elections							(0.475)
ControlsXXXXXXXFixed effectsCountryXXXXXXDisaster typeXXXXXX								
Fixed effectsCountryXXXXXXDisaster typeXXXXXXX	Controls	Х	Х	Х	Х	Х	Х	Х
Fixed effectsCountryXXXXXDisaster typeXXXXXX								
CountryXXXXXXDisaster typeXXXXXXX	Fixed effects							
Disaster type X X X X X X X X	Country	Х	Х	Х	Х	Х	Х	Х
	Disaster type	Х	Х	Х	Х	Х	Х	Х
Year X X X X X X X X	Year	Х	Х	Х	Х	Х	Х	Х
Observations 761 836 551 551 835 835 835	Observations	761	836	551	551	835	835	835
R^2 0.402 0.348 0.389 0.378 0.345 0.344 0.344	R^2	0.402	0.348	0.389	0.378	0.345	0.344	0.344

Table 7. Birth region effect and factors influencing US evaluation of appeals

Notes: ***, ** denote statistical significance at the 1, 5 and 10% level respectively, with clustered standard errors in parentheses. The dependent variable is the disaster-specific log funding in 2017 US dollars, conditioning on having received any funding. The variable birth region takes the value of one for disasters that hit the birth region of the country leader and zero otherwise. The main effects of the interacted variables are included in the regressions.

³⁹ A drawback of the ICRG data is that they are only available to us for the sub-period 1985 to 2012 and fully unavailable for Afghanistan, Nepal and Tajikistan.

⁴⁰ Results are unchanged when we focus on 11 months after the disaster occurs.

Table 7 reports the additional results. According to column 1, US media attention does not significantly moderate the degree of regional favoritism. As shown in column 2, countries with a higher fraction of public relative to particularistic spending behavior exhibit less regional favoritism. Moreover, both a better quality of the local bureaucracy (column 3) as well as higher government accountability (column 4) are significantly associated with less regional favoritism. Columns 5 to 7 show that none of the interactions with election indicators is statistically significant. Taken together, the evidence is thus not consistent with a storyline in which the United States carefully polices its funds with the goal to prevent potential abuse for domestic political purposes. Quite the contrary, it seems that recipient country leaders prone to clientelism and with weak bureaucracy and governance channel higher amounts of aid to disasters hitting their birth regions.

Our results also speak to potential motives of recipient country leaders. Conceptually, altruistic, electoral or political reasons more broadly might drive the home bias that we detect. A detailed test of these different channels is beyond the means of this paper. However, our results show that short-term electoral motives do not seem to dominate, as favoritism should then become stronger in the run-up for national elections in disaster-affected countries. As we do not find evidence for electoral motives behind the observed birth region effect, favoritism is more likely to reflect leaders' intrinsic desire to assist their own birth regions.⁴¹ We find the home bias to be stronger rather than weaker in countries with stronger clientelism and weaker bureaucracy as well as governance, which are arguably environments that give domestic political leaders more leeway in pursuing their own interests. To the contrary, we have no reason to assume that the US government would want to channel more aid to the birth regions of the leaders of such countries. In summary, it thus seems that OFDA either lacks the motive or the means to prevent recipient governments from favoring birth regions, but does not actively support the misallocation of funds.

⁴¹ This interpretation matches evidence from Vietnam according to which politicians channel public resources to their hometowns due to social preferences rather than expected political gains (Do et al., 2017).

7. Extensions

Total US disaster assistance

As previously discussed in Section 2, OFDA is the chief agency for US disaster assistance but there are other USAID offices (e.g., Office of Food for Peace) or ministries that also contribute humanitarian aid for certain disasters. Given that OFDA provides approximate numbers for these contributions for the 1964 to 2004 sub-period, we are able to formally test whether the previously obtained results are representative for the United States at large, or pertain to OFDA only.⁴² In Table 8, we replicate columns 1 to 3 of Table 4, using total US government contributions as reported by OFDA. To facilitate comparison, we also show regressions with OFDA funding restricted to the same sub-period. While the results for OFDA remain similar to those obtained with the full sample, point estimates increase in magnitude when we include all contributions, suggesting that favoritism might become more salient when multiple US agencies are involved.⁴³ Given these estimates, we interpret the key findings of this study to be representative for US humanitarian aid at large.

	(1)	(2)	(3)	(4)	(5)	(6)
	<i>All</i>	<i>All</i>	<i>All</i>	OFDA	OFDA	OFDA
Birth region	0.755 ^{***}	0.758 ^{***}	0.713 ^{***}	0.459 ^{***}	0.489 ^{***}	0.419 ^{**}
	(0.198)	(0.230)	(0.243)	(0.141)	(0.164)	(0.170)
Controls (local area characteristics)		Х	Х		Х	Х
Fixed effects Country Disaster type Year	Х	X X	X X X	Х	X X	X X X
Observations	633	633	633	587	587	587
R ²	0.171	0.241	0.318	0.149	0.235	0.313

Table 8. Total US government contributions, Log funding

Notes: ***, **, * denote statistical significance at the 1, 5 and 10% level respectively, with clustered standard errors in parentheses. The table replicates columns 1 to 3 from Table 4 using all US funding in columns 1 to 3 and OFDA only in columns 4 to 6.

⁴² Due to the shorter time period, within-country variation is reduced to the extent that we refrain from estimating disaster-area fixed effects models using total US government contributions.

⁴³ A potential reason for this could be that procedures become more complex, making it easier for a recipient country's political leaders to favor their regions.

Ethnic favoritism

A potential explanation for our main results relates to the overlap of birth regions with ethnic homelands of country leaders. As such, the estimated birth region effects might be driven by ethnicity rather than birth region. To test this possibility, we exploit variation in the ethnic identity of the population in disaster-affected areas and generate an alternative measure of favoritism based on the ethnic identity of the political leader.⁴⁴

To determine the geographic spread of ethnic settlement patterns, the political-economy literature on ethnicity heavily relies on ethno-linguistic or expert-based maps, such as *GREG* (Weidmann et al., 2010), Ethnologue (Gordon, 2005), and *GeoEPR* (Wucherpfennig et al., 2011).⁴⁵ In addition to being based on potentially outdated, one-dimensional, or incomplete information, these approaches come with the major disadvantage of failing to capture the nuanced realities of actual settlement patterns.⁴⁶ Therefore, this paper follows a novel approach of using IPUMS census data and DHS data.⁴⁷ IPUMS-International provides microdata for 98 countries. It currently covers more than one billion people in 443 censuses. The data are consistent across countries and over time and represent the largest available archive of publicly available census data (MEASURES DHS, 2017; Minnesota Population Center, 2017, 2019).

⁴⁴ For the identification of leader ethnicities, we augment the Archigos database with leader ethnicity information acquired from Fearon et al. (2007), Parks (2014) and through online search.

⁴⁵ See, for instance, Alesina et al. (2016), Guariso and Rogall (2017), and Anaxagorou et al. (2019).

⁴⁶ For instance, according to Weidmann et al. (2010), details on sources, definitions, and coding conventions of the *Atlas Narodov Mira* which serves as a base for GREG are not documented. Weidmann et al. (2010) however infer the coding criteria by comparing sub-samples with data on ethnicities from other sources. They conclude that the distinction between groups within countries is mainly based on language. This ignores important differences between ethnic groups. For example, the Sunni-Shi'ite division in Iraq is ignored, as is those between the Hutus and Tutsis in Rwanda, even though these are among the most important cleavages in their countries (Wucherpfennig et al., 2011).

⁴⁷ Weidmann et al. (2010: 492) mention the possibility to "infer the location of ethnic groups from survey or census data." According to Weidmann et al. (2010: 492) "providing spatially referenced census data for a larger set of cases is not possible." We disagree and do exactly this (see Gershman and Rivera, 2018, for a similar approach to coding the ethno*linguistic* composition of sub-national regions in Sub-Saharan Africa).

	(1)	(2)	(3)	(4)	(5)	(6)
	Log lunaing	Log lunaing	Log lunaing	Log lunaing	Log lunaing	Log lunaing
Birth region	0.492***	0.410**	0.362*	0.539***	0.431***	0.373**
C C	(0.159)	(0.161)	(0.180)	(0.149)	(0.150)	(0.163)
Ethnic homeland	0.117 [´]	0.085	0.069 [´]	(<i>'</i>	(<i>'</i>	(<i>'</i>
	(0.225)	(0.244)	(0.256)			
Ethnic share	、	()	(<i>,</i>	-0.046	0.062	0.238
				(0.303)	(0.341)	(0.387)
				、	x <i>y</i>	x <i>y</i>
Controls		Х	Х		Х	Х
Fixed effects						
Country	Х	Х	Х	Х	Х	Х
Disaster type		Х	Х		Х	Х
Year			Х			Х
Observations	673	673	673	673	673	673
R^2	0.137	0.224	0.339	0.136	0.223	0.339

Table 9. Birth region and ethnic homelands

Notes: Table replicates columns 1 to 3 from Table 4. ***, **, * denote statistical significance at the 1, 5 and 10% level respectively, with clustered standard errors in parentheses.

We exploit the precise estimates provided by the census data to define ADM1-level ethnic homelands satisfying at least one of the three following requirements: (i) the ethnic share of the region's population is of at least 90%, (ii) the region is among the top decile of ethnic shares, (iii) the ethnic share of the region's population is between the highest ethnic share observed in other regions and 10% below. Moreover, in cases where (ii) holds but the ethnic share of the region's population is less than 50% of the highest ethnic share observed in other regions, we do not consider the region an ethnic homeland.⁴⁸ Once all ethnic homelands are determined, we match them to the ethnicities of country leaders. We then code a binary indicator that equals one if a disaster hit the ethnic homeland of the political leader.⁴⁹ To make sure that our results do not depend on specific choices made when defining ethnic homelands, we further create a continuous measure capturing

⁴⁸ Rule (i) describes ethnic majority regions. Rule (ii) is useful to identify homelands of ethnic minorities who never reach a population share of 90%. We introduce rule (iii) as it may be the case that regions that are covered by rule (ii) do not necessarily exhibit a substantially higher population share than those closely below this cutoff. It is further possible that regions covered by rule (ii) comprise both areas with very high and very low population shares if an ethnicity is mainly centered in one or two regions of a country. The final requirement therefore imposes a restriction on the allowed divergence from the most important region from the perspective of the ethnicity.

⁴⁹ Note that given the definition of ethnic homelands applied in this paper, it is possible that a leader has multiple ethnic homelands. In these cases, the binary indicator equals one if a disaster hit at least one of these homelands.

the share of the population in disaster-affected areas belonging to the same ethnicity as the political leader ("ethnic share").⁵⁰

In Table 9, we replicate columns 1 to 3 of Table 4 adding either the binary ethnic homeland indicator or the ethnic share. While the sample is reduced to 37 countries for which ethnic data are available, the birth region effect remains both economically and statistically significant in all specifications. By contrast, the influence of both ethnic homelands and ethnic share never reach statistical significance. Consequently, ethnic power relations do not seem to explain the observed birth region effect.⁵¹

8. Conclusion

With poor response capacities and vulnerable economies, low- and middle-income countries are particularly exposed to the impacts of natural disasters. Over the 1964 to 2017 period, each of such calamities in our sample resulted in the death of 360 people, on average. In addition to high casualty numbers, natural disasters can potentially lead to a range of adverse outcomes such as weaker economic growth (Felbermayr and Gröschl, 2014) or educational deficits of future generations (Caruso, 2017). International humanitarian aid therefore potentially fulfills a crucial role by providing affected countries with much-needed relief goods, financial means, organizational support and expertise, and thereby reducing the macro- and micro-costs of natural disasters.

However, in order to ensure that aid is granted to those who are most in need, it is essential to understand political and economic rationales that could distort the allocation

⁵⁰ Given that the geographic distribution of disaster victims is not reported, disasters affecting multiple regions are handled by calculating an area-weighted average of the ethnic composition of all affected ADM1 regions (or ADM2 areas if available).

⁵¹ As Ahlerup and Isaksson (2015) point out, ethnic and regional favoritism are conceptually distinct from each other. Our results match those of Dreher et al. (2019) for Chinese development finance. However, unlike Dreher et al., who analyze ethnicity effects at the level of traditionally defined ethnic homelands based on *GREG* data, we are able to analyze both birth region and ethnicity effects at the level of ADM1 regions for aid flows directed to the same regions, ruling out the possibility that the differential effects are driven by varying geographic coverage. This feature of our data further allows us to estimate interaction effects between ethnic homelands and birth regions. These are, however, statistically insignificant in all specifications (results available on request).

of aid and thus reduce the effectiveness of disaster relief.⁵² Our study contributes to the literature by taking a unique and new perspective that puts a *domestic political factor within a large number of recipient countries* at its center. In the first sub-national analysis of its kind, we have uncovered the importance of regional favoritism for the allocation of disaster aid. According to our results, disasters hitting birth regions of political leaders receive substantially higher amounts of humanitarian aid than other comparable disasters. This result is robust to netting out effects of local area characteristics and disaster magnitude.

Moreover, both disaster-area fixed effects and placebo models indicate that the effect is not related to the inherent unobserved characteristics of the affected areas, but attributed to the importance they hold when being hit by disasters. Furthermore, we do not find evidence that ethnic power relations govern the observed relationship. Nor is there evidence that the United States government actively promotes the birth region effect in order to please political leaders of countries with close political or commercial ties. To the contrary, regional favoritism seems to be weaker in countries where social and infrastructure spending tends to follow public interests, where bureaucracies are more able to resist political influence, and where governments are held accountable by reliable institutions and political competition, suggesting that the birth-region effect is driven by recipient country politics, facilitated by US policies that seem to neglect the internal political dynamics in the countries receiving its aid.

Our results have important policy implications. First, we identify the existence of favoritism in an extreme scenario of humanitarian need. While similar forms of favoritism have previously been observed in the contexts of Chinese development aid (Dreher et al., 2019) and economic growth (Hodler and Raschky, 2014), the fact that discriminatory politics are applied even when human lives are directly at stake is alarming. Our analysis thus calls for further research on this topic and the exploration of realistic possibilities of engaging in corrective action. In particular, even though we do not find evidence that the observed effects are driven by donor's political and economic interests, it is imperative to revise

⁵² On the importance of political motives for the effectiveness of aid, see Dreher et al. (2018).

established processes related to disaster damage assessment and aid allocation in order to reduce the scope for manipulation by recipient country leaders.

This policy implication should be prioritized for two reasons. First, given that the results appear to be driven by recipient countries, it is likely that aid from other donor countries is similarly affected to the extent that these donors apply decision-making processes comparable to those of the United States. Second, as with accelerating climate change, natural disasters are projected to significantly increase in both frequency and intensity (IPCC, 2018), the relevance of effective humanitarian aid will likely increase in the near future.

A further policy implication of our findings relates to the long-term consequences of natural disasters. As leaders' birth regions tend to already be among the richer regions of their country (Hodler and Raschky, 2014; Dreher et al., 2019), regional favoritism in humanitarian aid could further exacerbate within-country inequalities. This is especially problematic given the potential of natural disasters to cause adverse long-run social and economic consequences. In order to ensure that disaster relief activities do not counteract international efforts to reduce inequality within poor countries, donors need to increase their efforts to incorporate concerns of regional equity in their aid allocation decisions.

References

- Ahlerup, P. and Isaksson, A.-S. (2015). Ethno-Regional Favouritism in Sub-Saharan Africa. *Kyklos*, 68(2), 143–152.
- Alesina, A., and Dollar, D. (2000). Who gives Foreign Aid to Whom and Why? *Journal of Economic Growth*, 5(1), 33–63.
- Alesina, A., Michalopoulos, S., and Papaioannou, E. (2016). Ethnic Inequality. *Journal of Political Economy*, 124(2), 428–488.
- Anaxagorou, C., Efthyvoulou, G., and Sarantides, V. (2019). Re-election Incentives, Ethnic Favoritism and the Subnational Allocation of Foreign Aid. Paper presented at the EEA-ESEM 2019. Available at: https://editorialexpress.com/cgibin/conference/download.cgi?db_name=EEAESEM2019&paper_id=1230 [accessed August 7, 2019].
- Annen, K., and Strickland, S. (2017). Global Samaritans? Donor Election Cycles and the Allocation of Humanitarian Aid. *European Economic Review*, *96*, 38–47.
- Arino, O., Ramos, J., Kalogirou, V., Bontemps, S., Defourny, P., and Van Bogaert, E. (2012). Global Land Cover Map for 2009. European Space Agency (ESA) & Université Catholique de Louvain (UCL), PANGAEA.
- Archigos (2018). A Data Base on Leaders 1875-2004. Available at: http://www.rochester.edu/college/faculty/hgoemans/data.htm [accessed January 22, 2018].
- Ball, R., and Johnson, C. (1996). Political, Economic, and Humanitarian Motivations for PL 480 Food Aid: Evidence from Africa. *Economic Development and Cultural Change*, 44(3), 515–537.
- Barbieri, K., Keshk, O. M. G., and Pollins, B. M. (2009). Trading Data: Evaluating Our Assumptions and Coding Rules. *Conflict Management and Peace Science*, 26(5), 471–491.
- Barbieri, K., and Keshk, O. (2016). Correlates of War Project Trade Data Set Codebook. Available at: http://correlatesofwar.org [accessed March 13, 2018].
- Berger, D., Easterly, W., Nunn, N., and Satyanath, S. (2013). Commercial Imperialism? Political Influence and Trade During the Cold War. *American Economic Review*, 103(2), 863–96.
- Bolin, B., and Kurtz, L. C. (2018). Race, Class, Ethnicity, and Disaster Vulnerability. In *Handbook of Disaster Research. Handbooks of Sociology and Social Research*, edited by Rodriguez, H., Donner W., and Trainor J. Springer. 181–203.

- Brazys, S., Elkink, J.A., and Kelly, G. (2017). Bad Neighbors? How Co-located Chinese and World Bank Development Projects Impact Local Corruption in Tanzania. *Review* of International Organizations 12: 227–253.
- Broussard, N. H., Dercon, S., and Somanathan, R. (2014). Aid and Agency in Africa: Explaining Food Disbursements across Ethiopian Households, 1994-2004. *Journal* of Development Economics, 108, 128–137.
- Caruso, G. (2017). The Legacy of Natural Disasters: The Intergenerational Impact of 100 Years of Disasters in Latin America. *Journal of Development Economics*, 127, 209– 233.
- Coppedge, M., Gerring, J., Knutsen, C., Lindberg, S., Skaaning, S., Teorell, J., Altman, D., Bernhard, M., Cornell, A., Fish, M., Gjerløw, H., Glynn, A., Hicken, A., Krusell, J., Lührmann, A., Marquardt, K., McMann, K., Mechkova, V., Olin, M., Paxton, P., Pemstein, D., Seim, B., Sigman, R., Staton, J., Sundström, A., Tzelgov, E., Uberti, L., Wang, Y., Wig, T., and Ziblatt, D. (2018). V-Dem Codebook v8. Varieties of Democracy (V-Dem) Project.
- CIESIN (2017). Center for International Earth Science Information Network. Columbia University. Gridded Population of the World, Version 4 (GPWv4): Population Count, Revision 10. Palisades, NY: NASA Socioeconomic Data and Applications Center (SEDAC). Available at: https://doi.org/10.7927/H4PG1PPM [accessed March 20, 2018]
- Cruz, C., Keefer, P. and Scartascini, C. (2018). Database of Political Institutions 2017 (DPI2017). Inter-American Development Bank. Numbers for Development. Available at: https://mydata.iadb.org/Reform-Modernization-of-the-State/Database-of-Political-Institutions-2017/938i-s2bw [accessed May 2, 2019]
- De Luca, G., Hodler, R., Raschky, P. A., and Valsecchi, M. (2018). Ethnic Favoritism: An Axiom of Politics? *Journal of Development Economics*, 132, 115–129.
- Dionne, K. Y., Kramon, E., and Roberts, T. (2013). Aid Effectiveness and Allocation: Evidence from Malawi. Paper presented at the Conference Research Frontiers in Foreign Aid at Princeton University, April 26-27.
- Do, Q.-A., Nguyen, K.-T., and Tran, A.N. (2017). One Mandarin Benefits the Whole Clan: Hometown Favoritism in an Authoritarian Regime. *American Economic Journal: Applied Economics*, 9(4), 1–29.
- Dreher, A., Eichenauer, V., and Gehring, K. (2018). Geopolitics, Aid and Growth: The Impact of UN Security Council Membership on the Effectiveness of Aid. *World Bank Economic Review*, 32: 268–286.

- Dreher, A., Fuchs, A., Hodler, R., Parks, B., Raschky, P., and Tierney, M. (2019). *African Leaders and the Geography of China's Foreign Assistance. Journal of Development Economics*, 140, 44–71.
- Drury, A. C., Olson, R. S., and Van Belle, D. A. (2005). The Politics of Humanitarian Aid: US Foreign Disaster Assistance, 1964-1995. *Journal of Politics*, 67(2), 454–473.
- Eichenauer, V., Fuchs, A., Kunze, S., and Strobl, E. (2019). Distortions in Aid Allocation of United Nations Flash Appeals: Evidence from the 2015 Nepal Earthquake, Kiel Working Paper No. 2120.
- Eisensee, T., and Strömberg, D. (2007). News Droughts, News floods, and US Disaster Relief. *Quarterly Journal of Economics*, 122(2), 693–728.
- EM-DAT (2018a). Frequently Asked Questions. Available at: http://www.emdat.be/frequently-asked-questions [accessed January 12, 2018].
- EM-DAT (2018b). Database Guidelines. Available at: https://www.emdat.be/guidelines [accessed January 12, 2018].
- Faye, M., and Niehaus, P. (2012). Political Aid Cycles. *American Economic Review* 102(7), 3516–3530.
- Fearon, J., Kasara, K., and Laitin, D. (2007). Ethnic Minority Rule and Civil War Onset. *American Political Science Review*, 101(1), 187–193.
- Felbermayr, G., and Gröschl, J. (2014). Naturally Negative: The Growth Effects of Natural Disasters. *Journal of Development Economics*, 111, 92–106.
- Fink, G., and Redaelli, S. (2011). Determinants of International Emergency Aid Humanitarian Need Only? *World Development*, 39(5), 741–757.
- Francken, N., Minten, B., and Swinnen, J. F. M. (2012). The Political Economy of Relief Aid Allocation: Evidence from Madagascar. *World Development*, 40(3), 486–500.
- Fuchs, A., and Klann, N.-H. (2013). Paying a Visit: The Dalai Lama Effect on International Trade. *Journal of International Economics* 91(1), 164–177.
- Fuchs, A., and Klann, N.-H. (2012). Emergency aid 2.0. In *Political Determinants of Foreign Aid and International Trade of Emerging Economies*, Doctoral dissertation, University of Goettingen.
- Gershman, B., and Rivera, D. (2018). Subnational Diversity in Sub-Saharan Africa: Insights from a New Dataset. *Journal of Development Economics*, 133, 231–263.
- Goemans, H. E., Gleditsch, K. S., and Chiozza, G. (2009). Introducing Archigos: A Dataset of Political Leaders. *Journal of Peace Research*, 46(2), 269–283.

Gordon, R. (2005). Ethnologue: Languages of the World. Sil International. Dallas, TX.

- Guariso, A. and Rogall, T. (2017). Rainfall Inequality, Political Power, and Ethnic Conflict in Africa. LICOS Discussion Paper 391/2017.
- Guha-Sapir, D., Below, R., Hoyois, R., and Hoyois, P. (2018). EM-DAT: The CRED/OFDA International Disaster Database [dataset]. Brussels, Belgium: Université Catholique de Louvain.
- Harmer, A., and Cotterrell, L. (2005). *Diversity in Donorship: The Changing Landscape of Official Humanitarian Aid.* Edited by Humanitarian Policy Group. Overseas Development. Humanitarian Policy Group. Overseas Development Institute. London, UK.
- Hodler, R., and Raschky, P. A. (2014). Regional Favoritism. *Quarterly Journal of Economics*, 129(2), 995–1033.
- Hoeffler, A., and Outram, V. (2011). Need, Merit, or Self-Interest What Determines the Allocation of Aid? *Review of Development Economics*, 15(2), 237–250.
- ICRG (2013). International Country Risk Guide (ICRG) Researchers Dataset, https://doi.org/10.7910/DVN/4YHTPU, Harvard Dataverse, V6, UNF:6:ay+aVDktpIU/TpZxygRE/Q== [fileUNF] [accessed August 24, 2017]
- IPCC (2018). Intergovernmental Panel on Climate Change. Global Warming of 1.5°C. An IPCC Special Report on the impacts of global warming of 1.5°C above pre-industrial levels and related global greenhouse gas emission pathways, in the context of strengthening the global response to the threat of climate change, sustainable development, and efforts to eradicate poverty. [Masson-Delmotte, V., P. Zhai, H.-O. Pörtner, D. Roberts, J. Skea, P.R. Shukla, A. Pirani, W. Moufouma-Okia, C. Péan, R. Pidcock, S. Connors, J.B.R. Matthews, Y. Chen, X. Zhou, M.I. Gomis, E. Lonnoy, T. Maycock, M. Tignor, and T. Waterfield (eds.)].
- International Dalit Solidarity Network. (2013). *Equality in Aid: Addressing Caste Discrimination in Humanitarian Response*. Copenhagen, Denmark.
- Jablonski, R. S. (2014). How Aid Targets Votes: The Impact of Electoral Incentives on Foreign Aid Distribution. *World Politics*, 66(2), 293–330.
- Jayne, T. S., Strauss, J., Yamano, T., and Molla, D. (2002). Targeting of Food Aid in Rural Ethiopia: Chronic Need or Inertia? *Journal of Development Economics*, 68(2), 247–288.
- Karbownik, K, and Wray, A. (2019). Long Run Consequences of Exposure to Natural Disasters. *Journal of Labor Economics*, 37(3), 949–1007.

- Kevlihan, R., DeRouen Jr, K., and Biglaiser, G. (2014). Is US Humanitarian Aid Based Primarily on Need or Self-interest? *International Studies Quarterly*, 58(4), 839–854.
- Klynman, Y., Kouppari, N., and Mukhier, M. (2007). *World Disasters Report: Focus on Discrimination*. Geneve, Switzerland: International Federation of Red Cross & Red Crescent Societies.
- Kunze, S., and S. Schneider (2019). *Disastrous Discretion The Nonlinear Political Bias in U.S Hurricane Relief.* ETH Zurich, Switzerland.
- Kuziemko, I., and Werker, E. (2006). How Much Is a Seat on the Security Council Worth? Foreign Aid and Bribery at the United Nations. *Journal of Political Economy*, 114(5), 905–930.
- Margesson, R. (2013). International Crises and Disasters: U.S. Humanitarian Assistance Response Mechanisms. Congressional Research Service Report for Congress. Washington D.C.
- Martin, M., and Margesson, R. (2008). Cyclone Nargis and Burma's Constitutional Referendum. Congressional Research Service Report for Congress. Washington D.C.
- Masaki, T. (2018). The Political Economy of Aid Allocation in Africa: Evidence from Zambia. *African Studies Review*, 61(1), 55–82.
- MEASURES DHS (2017). Demographic and Health Surveys. Available at: http://www.measuredhs.com/ [accessed March 7, 2018]
- Minnesota Population Center (2017). Integrated Public Use Microdata Series, International: Version 6.5 [dataset].
- Minnesota Population Center (2019). Integrated Public Use Microdata Series, International Website. Available at: https://international.ipums.org/international/ [accessed May 28 and August 15, 2019]
- NGIA (2017). US National Geospatial Intelligence Agency. World Port Index. Twenty sixth edition (Pub150).
- NOAA (n.d.). Version 4 DMSP-OLS Nighttime Lights Time Series. DMSP data collected by US Air Force Weather Agency. Available at: https://ngdc.noaa.gov/eog/dmsp/downloadV4composites.html [accessed January 9, 2018].
- Nunn, N., and Puga, D. (2012). Ruggedness: The Blessing of Bad Geography in Africa. *Review of Economics and Statistics*, 94(1), 20–36.

- Nunnenkamp, P., Öhler, H., Sosa Andrés, M. (2017). Need, Merit and Politics in Multilateral Aid Allocation: A District-level Analysis of World Bank Projects in India. *Review of Development Economics* 21(1): 126–156.
- OECD (2019). Humanitarian Assistance. Online databases. Humanitarian Assistance by DAC donor. Available at: http://www.oecd.org/dac/stats/humanitarian-assistance.htm [accessed February 20, 2019].
- OFDA (1990-2017). Annual Reports. Washington DC. Available at: https://dec.usaid.gov/dec/search/SearchResults.aspx?q=ZG9jdW1lbnRzLndlYl9jb2x sZWN0aW9uPSJvZmRhIGFubnVhbCByZXBvcnQi [accessed January 15, 2018].
- Parks, B. C. (2014). Brokering Development Policy Change: The Parallel Pursuit of Millennium Challenge Account Resources and Reform. Ph.D. Dissertation. London, UK: London School of Economics and Political Science.
- Persson, G. (2004). The Government's Humanitarian Aid Policy. Government Communication 2004/05: 52. Available at: https://www.government.se/49b74b/contentassets/df42b91831e34892b7d8e6ffea6f 19e5/the-governments-humanitarian-aid-policy [accessed August 19, 2019].
- Raschky, P. A., and Schwindt, M. (2012). On the Channel and Type of Aid: The Case of International Disaster Assistance. *European Journal of Political Economy*, 28(1), 119–131.
- Riley, S., Degloria, S., and Elliot, S. D. (1999). A Terrain Ruggedness Index that Quantifies Topographic Heterogeneity. *International Journal of Science*, 5, 23–27.
- Rommel, T., and P. Schaudt (2017). First Impressions: How Leader Changes Affect Bilateral Aid. CESifo Working Paper No. 6047.
- Roodman, D., Mackinnon, J.G., Ørregaard Nielsen, M. and Webb, M.D. (2019). Fast and wild: Bootstrap inference in Stata using boottest. *The Stata Journal*, 19(1), 4–60.
- Strömberg, D. (2007). Natural Disasters, Economic Development, and Humanitarian Aid. *Journal of Economic Perspectives*, 21(3), 199–222.
- Tatem, A. (2017). WorldPop, Open Data for Spatial Demography. *Nature, Scientific Data*, 4(170004).
- USAID (2005). ADS chapter 251: International Disaster Assistance. USAID.
- USAID (2018a). Office of U.S. Disaster Foreign Disaster Assistance. Available at: https://www.usaid.gov/who-we-are/organization/bureaus/bureau-democracyconflict-and-humanitarian-assistance/office-us [accessed January 5, 2018].

- USAID (2018b). OFDA Reports Archive. Available at: https://dec.usaid.gov/dec/search/SearchResults.aspx?q=ZG9jdW1lbnRzLndlYl9jb2x sZWN0aW9uPSJvZmRhIGFubnVhbCByZXBvcnQi [accessed January 15, 2018].
- UN (2018). World Urbanization Prospects: The 2018 Revision. United Nations, Department of Economic and Social Affairs, Population Division. Working Paper No. ESA/P/WP 252. New York: United Nations.
- Vanderbilt Television News Archive (n.d.). Nashville, TN: Vanderbilt University. Available at: http://tvnews.vanderbilt.edu.
- Voeten, E. (2013). Data and Analyses of Voting in the UN General Assembly. Routledge Handbook of International Organization, edited by Bob Reinalda. Available at SSRN: http://ssrn.com/abstract=2111149 [dataset].
- Vreeland, J., and Dreher, A. (2014). *The Political Economy of the United Nations Security Council. Money and Influence*. Cambridge University Press.
- Weidmann, N. B., Rød, J. K., and Cederman, L.-E. (2010). Representing Ethnic Groups in Space: A New Dataset. *Journal of Peace Research*, 47(4), 491–499.
- WNA (2017). World Nuclear Association. Country Profiles. Available at: http://www.worldnuclear.org/information-library/country-profiles.aspx [accessed January 20, 2019].
- Wucherpfennig, J., Weidmann, N. B., Girardin, L., Cederman, L.-E., and Wimmer, A. (2011). Politically Relevant Ethnic Groups across Space and Time: Introducing the GeoEPR Dataset. *Conflict Management and Peace Science*, 28(5), 423–437.

Appendix

$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Country	Barren land (%)	# ports	# nuclear plants	Capital city (%)	Night light per 1000 people ^a	Population densitiy ^b	City population ^b	Ruggedness ^c
(16.5) (0.0) (0.0) (1.8) (1.0) (22.3) (2.2) Angoh (0.2,7) (0.0,0) (0.0,0,0) (1.8) (1.0) (0.0,1,1) (0.0,1,2) (0.1,1) Angoh (0.2,2) (0.0,0,1) (0.0,2,1) (0.0,1,2) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1) (0.1,1)	Afghanistan	35.4	0.0	0.0	10.0	1.3	0.3	360.6	3.7
(b 2,706) (b 0,0.0) (b 0,0.3) (b 0,4.720.9) (b 1,4.720.9) Argenina 0.2 0.00 0.00 (b 1,4.83) 0.5 2517.7 0.0 Argenina 5.1 0.2 0.6.00 (b 1,4.86) (b 1,4.86) 0.1 3740.0 0.0 Argenina (7.4) (0.4) (0.6) (b 0,5.66) (b 0,0.10) (b 0,5.66) (b 0,0.10) (b 0,5.66) (b 0,5.66) <td>•</td> <td>(15.6)</td> <td>(0.0)</td> <td>(0.0)</td> <td></td> <td>(1.8)</td> <td>(1.0)</td> <td>(923.0)</td> <td>(2.0)</td>	•	(15.6)	(0.0)	(0.0)		(1.8)	(1.0)	(923.0)	(2.0)
Angela 0.2 0.0 0.0 41.8 3.9 0.5 2516.7 0 Argenina (0.2, 2) (0.0, 0) (0.0, 0) (0.2, 1) (0.2, 10) (0.1, 12) (0.2, 10) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1, 12) (0.1,		[0.2,70.6]	[0.0,0.0]	[0.0,0.0]		[0.0,8.3]	[0.0,6.1]	[0.0,4720.9]	[0.1,8.6]
(0.4) (0.0) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) (0.1) <th< td=""><td>Angola</td><td>0.2</td><td>0.0</td><td>0.0</td><td>41.8</td><td>3.9</td><td>0.5</td><td>2516.7</td><td>0.8</td></th<>	Angola	0.2	0.0	0.0	41.8	3.9	0.5	2516.7	0.8
Internation (b)		(0.4)	(0.0)	(0.0)		(2.6)	(1.3)	(2939.2)	(0.5)
Augement Cr.J. Co.J. Co.J. <thco.j.< th=""> Co.J. Co.J. <</thco.j.<>	A	[0.0,2.2]	[0.0,0.0]	[0.0,0.0]	45.4	[0.2,11.8]	[0.0,6.5]	[0.0,10612.5]	[0.1,3.0]
0.0 4 3 eg 0.0 1 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0	Argentina	5.1	0.2	0.6	15.1	86.0	0.1	3740.0	0.6
Bangladesh (b) 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0,		(7.4) [0.0.26.6]	(0.4)	(0.0)		(40.0)	(0.4)	(0.0 22960 41	(0.9)
Observation (b) 5 (b) 0 (c) 0	Bandladesh	0.3	[0.0, 1.0]	[0.0,3.0]	38.1	[19.9,207.2] 2.4	[0.0,3.0]	[0.0,23800.4] 6643 3	0.1
(b) 0.2 / 3] (b) 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0 / 0.0	Darigiadesii	(0.5)	(0,0)	(0.0)	50.1	(0.8)	(0.9)	(6882.6)	(0.2)
Bolivia (4.4) (0.0) (0.0) (0.0) (1.6) (0.0) (1.1) (1.1) (0.55,56] (0.0,0.0) (0.0,0.0) (5.4,52,5] (0.0,0.5] (0.0,4599,8] (0.1) (0.3) (0.7) (0.7) (1.7,75,2] (2.1) (1.2,854,81) (0.0,0.5) Chile (1.6,7) (0.0,0.0) (1.4,422,81) (0.0,0.2,22,21,22,22,23,23,23,23,23,23,23,23,23,23,23,		[0.0.2.9]	[0.0.0]	[0.0.0]		[0.3.4.6]	[0.1.8.2]	[108.0.27684.1]	[0.0.1.3]
(B,0) (0,0) (0,0) (0,0) (B,0) (D,1) (H455,0) (D,1) Brazi 0.2 0.6 0.3 0.6 43.5 1.0 100.459,03 [0,1,4] (D,1) (D,1) <td>Bolivia</td> <td>4.4</td> <td>0.0</td> <td>0.0</td> <td>56.7</td> <td>18.6</td> <td>0.0</td> <td>1811.9</td> <td>1.2</td>	Bolivia	4.4	0.0	0.0	56.7	18.6	0.0	1811.9	1.2
[0.6,55.6] [0.0.0] [0.0.0] [5.4,52.5] [0.0.0.5] [0.4,59.8] [0.1] max (0.3) (0.7) (0.7) (17.8) (2.1) (12518.2) (0.0) Chile 16.4 0.0 0.0 36.4 663 0.2 22628.6 21 Chile 16.4 0.0 0.0 36.4 663 0.0 22628.6 21 Chile 16.4 0.0 0.0 36.4 663 0.0 22628.6 21 Chile 16.0 0.0 14.4 2.4 13.0 0.3 2269.6 12 Chile 1.4 1.4 1.4 1.4 1.4 1.4 1.6 601.8 11 1.6 Chile 0.0 0.0 0.0 1.6 1.6 1.6 1.6 1.6 1.6 1.6 1.6 1.6 1.6 1.6 1.6 1.6 1.6 1.6 1.6 1.6 1.6 1.6 1.6		(8.0)	(0.0)	(0.0)		(8.0)	(0.1)	(1495.0)	(1.0)
Brazi 0.2 0.6 0.3 0.6 4.3.5 1.0 10548.3 0.1 C (0.0.2.7] (0.0.3.0] (0.0.7) (7.7.75.2] (0.0.8.4] (19.4,459.21.8] (0.0.7) Chile (2.7.7) (0.0.0) (0.0.4) (7.1.8) (0.2.7) (0.0.8.7) (11.7.8) Chile (19.4) (0.0.1) (0.0.4) (14.4.8.9.9) (0.2.7) (0.0.8.7) (11.7.4.7.8) (0.1.7.8.7.8.7.8.7.8.7.8.7.8.7.8.7.8.7.8.7		[0.0,55.6]	[0.0,0.0]	[0.0,0.0]		[5.4,52.5]	[0.0,0.5]	[0.0,4599.8]	[0.1,4.7]
	Brazil	0.2	0.6	0.3	0.6	43.5	1.0	10548.3	0.5
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $		(0.3)	(0.7)	(0.7)		(17.8)	(2.1)	(12518.2)	(0.4)
Chile 16.4 0.0 0.0 36.4 69.3 0.2 2628.6 2.4 [0.0,76.6] [0.0,00] [0.0,00] [14.4,423) [0.0,20] [0.0,39115.4] [1.0] China 8.1 0.1 1.4 2.4 [1.3.0] [0.3.39115.7] [1.0] Colombia 0.3 [0.0.40] [0.0.22.0] [2.2,70.5] [0.0.5.9] [0.6481157.6] [0.0.20] Colombia 0.5 (0.0) (0.0) [0.0.0.7] [0.0.23161.7] [0.0.2] [1.1] [0.0.4.7] [0.0.0.0] [0.0.0.0] [0.0.23161.7] [0.0.23161.7] [0.0.23161.7] [0.0.23161.7] [0.0.23161.7] [0.0.23161.7] [0.0.23161.7] [0.0.23161.7] [0.0.23161.7] [0.0.23161.7] [0.0.23161.7] [0.0.23161.7] [0.0.22161.7] [0.0.22161.7] [0.0.22161.7] [0.0.22161.7] [0.0.22161.7] [0.0.22161.7] [0.0.22161.7] [0.0.22161.7] [0.0.22161.7] [0.0.22161.7] [0.0.22161.7] [0.0.22161.7] [0.0.22161.7] [0.0.22161.7] [0.0.201 [0.0.201.7] [0.		[0.0,2.7]	[0.0,3.0]	[0.0,2.0]		[7.7,75.2]	[0.0,8.4]	[89.4,59021.8]	[0.0,2.9]
(21,7) (0,0) (0,0) (71.8) (0.5) (3112.3) (1.1) China 8.1 0.1 1.4 2.4 13.0 0.3 338812 2.2 (19.4) (0.4) (3.4) (0.4) (0.5.9) (0.0.4711576) (0.5.1) Colombia 0.0 0.0 16.1 (2.7.6) (0.5.9) (0.0.47117 (0.0.1) Colombia 0.0 0.0 0.0 16.1 (9.9.74) (0.4) (0.47117 (0.0.1) (0.0.21161717 (0.0.1) (0.0.21161717 (0.0.1) (0.0.21161717 (0.0.1) (0.0.21161717 (0.0.1) (0.0.21161717 (0.0.1) (0.0.21161717 (0.0.1) (0.0.21161717 (0.0.1) (0.0.21161717 (0.0.1) (0.0.21161717 (0.0.1) (0.0.21161717 (0.0.1) (0.0.116100071 (0.0.0) (11.2) (0.1) (0.0.211610077600 (11.0) (0.0.1) (0.0.11610077600 (11.0) (0.0.1) (0.0.1161007771 (0.0.1) (0.0.211610077600771 (0.0.1) (0.0.0,0) (0.0.0) (0.0	Chile	16.4	0.0	0.0	36.4	69.3	0.2	2628.6	2.8
		(21.7)	(0.0)	(0.0)		(71.8)	(0.5)	(3112.3)	(1.0)
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	China	[0.0,78.6]	[0.0,0.0]	[0.0,0.0]	0.4	[14.4,429.9]	[0.0,2.0]	[0.0,9115.4]	[1.0,7.1]
(15.9) (10.4) (10.220) (12.705) (10.50) (10.407) (10.107) Colombia (0.5) (0.0) (0.0) (16.1) (2.5705) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747) (10.6747	China	8.1 (10.4)	0.1	1.4	2.4	13.0	0.3	32801.2	2.3
Colombia (b) A(x) (c) A(x) (c) A(x) (c) A(x) (c) A(x) (c) A(x)		(19.4) [0.0.88.1]	[0.4]	(3.4)		(9.4)	[0.05.0]	(00190.0) [0.0.481157.6]	[0 0 7 7]
Construct (0.5) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0)	Colombia	0.3	0.0	0.0	16.1	24.5	0.6	3754 1	1 7
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Colombia	(0.5)	(0.0)	(0.0)	10.1	(13.1)	(1.6)	(6210.2)	(1.0)
Costa Rica 0.0 '0.0 '0.0 '0.0 '0.0 '0.0 '1.2 '0.1 '47.8 '2.2 [0.0,0,0] [0.0,0,0] [0.0,0,0] [13.2] (0.1) (54.2) (0.1) Cuba 0.0 1.8 0.0 22.0 0.4 '707.1 (0.0) [0.0,0.3] [0.0,0.6,0] (0.0) (1.3) (957.4) (0.0) [0.0,0.3] [0.0,0.0] [0.0,0.3] [0.0,16086.6] [0.0,0.0] (1.3) (957.4) (0.0) [0.0,0.1] [0.0,0.0] [0.0,0.0] [0.0,2.296.6] [0.0,2.15.5] [0.0,16086.6] (0.0) [0.0,0.0] [0.0,0.0] [0.0,0.0] [0.0,2.296.6] [0.0,2.15.5] [0.0,16086.6] (0.0.2 [0.0,0.0] [0.0,0.0] [0.0,0.0] [11.4.46.0] [0.1,2.8] (0.0,16086.6] (0.0.2 [0.0,0.0] [0.0,0.0] [11.4.45.0] (0.1,2.8] (0.1,2.8] (0.1,2.8] (0.1,2.8] (0.1,2.8] (0.1,2.8] (0.1,2.8] (0.1,2.1,1.9) (0.2,1.1,1.9)		[0.0.4.7]	[0.0.0]	[0.0.0]		[0.9.74.7]	[0.0.7.4]	[0.0.23161.7]	[0.0.5.5]
	Costa Rica	0.0	0.0	0.0	40.0	59.5	0.1	467.8	2.4
		(0.0)	(0.0)	(0.0)		(11.2)	(0.1)	(540.2)	(0.8)
		[0.0,0.0]	[0.0,0.0]	[0.0,0.0]		[39.4,84.8]	[0.0,0.3]	[0.0,1760.0]	[1.0,4.3]
	Cuba	0.0	1.8	0.0	22.4	22.0	0.4	707.1	0.8
		(0.0)	(1.6)	(0.0)		(10.3)	(1.3)	(957.4)	(0.5)
$\begin{array}{cccccccccccccccccccccccccccccccccccc$		[0.0,0.3]	[0.0,6.0]	[0.0,0.0]		[3.7,72.7]	[0.0,8.7]	[0.0,2913.0]	[0.0,2.3]
	DR Congo	0.0	0.0	0.0	12.7	1.2	1.5	1493.2	0.7
		(0.0)	(0.0)	(0.0)		(3.0)	(4.3)	(2910.7)	(0.6)
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Dominican Dan	[0.0,0.1]	[0.0,0.0]	[0.0,0.0]	21.6	[0.0,29.6]	[0.0,21.5]	[0.0,16086.6]	[0.0,2.2]
	Dominican Rep.	(0.0)	(0,0)	(0.0)	31.0	21.0	0.5	(124.7	(0.6)
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		[0.0,0]	[0 0 0 0]	[0 0 0 0]		[11 4 46 0]	[0 1 2 8]	[0 0 3491 1]	[0 2 3 6]
Construction Construction<	Ecuador	22	0.0	0.0	23.4	40.9	0.2	931 7	1 7
	200000	(10.1)	(0.0)	(0.0)	2011	(56.5)	(0.2)	(1312.0)	(1.1)
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		[0.0,56.9]	[0.0,0.0]	[0.0,0.0]		[1.5,495.8]	[0.0,0.6]	[0.0,4520.7]	[0.1,4.4]
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	El Salvador	0.0	0.0	0.0	59.5	21.2	0.9	602.9	1.8
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		(0.0)	(0.0)	(0.0)		(3.4)	(1.4)	(520.8)	(0.4)
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$		[0.0,0.0]	[0.0,0.0]	[0.0,0.0]		[12.2,26.9]	[0.2,5.9]	[0.0,1096.5]	[1.2,3.7]
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Ethiopia	8.0	0.0	0.0	10.2	0.7	0.3	400.9	1.5
		(15.5)	(0.0)	(0.0)		(1.2)	(1.3)	(820.0)	(1.1)
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	_	[0.0,57.1]	[0.0,0.0]	[0.0,0.0]		[0.0,6.3]	[0.0,7.2]	[0.0,3317.9]	[0.1,6.2]
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	Greece	0.2	0.3	0.0	18.8	136.9	1.4	960.2	3.0
		(1.2)	(0.4)	(0.0)		(58.0)	(4.1)	(1446.1)	(1.3)
Guateminia 0.0 0.0 0.0 0.0 0.0 13.4 0.7 737.2 2. (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (0.0) (1.5) (1078.4) (1.7) Haiti 0.0 0.0 0.0 51.6 1.2 0.8 982.3 2.4 (0.0) (0.0) (0.0) (0.0) (0.0) (0.0,0) (1002.7) (0.6 (0.0,0.0] [0.0,0.0] [0.0,0.0] [0.0,3.8] [0.1,3.7] [0.0,2642.8] [1.1,4] Honduras 0.0 0.0 0.0 33.3 17.5 0.2 551.5 2.7 (0.0) (0.0) (0.0) (0.0) (0.0) (0.4) (0.3) (613.0) (0.5) (10.0) [0.0,0.0] [0.0,0.0] [0.0,0.0] [2.1,51.6] [0.0,0.8] [0.0,2069.0] [0.4,4] India 2.6 0.3 1.2 6.0 9.8 1.3 17358.1 1.2 [0.0,0.3459	Guatamala	[0.0,9.7]	[0.0,1.0]	[0.0,0.0]	22.0	[5.4,245.8]	[0.0,16.6]	[0.0,3981.7]	[0.6,6.0]
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	Gualemaia	(0,0)	(0.0)	(0,0)	33.0	(6.9)	(1.5)	(1078.4)	(1 1)
Haiti $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ $[0.0,0.0]$ <td></td> <td>[0 0 0]</td> <td>[0,0,0]</td> <td>[0 0 0 0]</td> <td></td> <td>[0 9 41 3]</td> <td>[0 0 7 1]</td> <td>[0 0 2738 3]</td> <td>[0 1 5 0]</td>		[0 0 0]	[0,0,0]	[0 0 0 0]		[0 9 41 3]	[0 0 7 1]	[0 0 2738 3]	[0 1 5 0]
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	Haiti	0.0	0.0	0.0	51.6	1.2	0.8	982.3	2.5
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		(0.0)	(0.0)	(0.0)		(0.9)	(1.0)	(1062.7)	(0.6)
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$		[0.0,0.0]	[0.0,0.0]	[0.0,0.0]		[0.0,3.8]	[0.1,3.7]	[0.0,2642.8]	[1.1,4.4]
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	Honduras	0.0	0.0	0.0	33.3	17.5	0.2	551.5	2.1
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		(0.0)	(0.0)	(0.0)		(8.4)	(0.3)	(613.0)	(0.9)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		[0.0,0.0]	[0.0,0.0]	[0.0,0.0]		[2.1,51.6]	[0.0,0.8]	[0.0,2069.0]	[0.4,4.8]
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	India	2.6	0.3	1.2	6.0	9.8	1.3	17358.1	1.3
		(5.3)	(0.5)	(2.3)		(18.2)	(4.0)	(21380.6)	(1.8)
Indonesia 0.0 0.3 0.0 9.2 6.5 1.0 3451.8 1.1 (0.0) (0.5) (0.0) (5.7) (2.7) (5762.2) (0.1 [0.0,0.0] [0.0,2.0] [0.0,0.0] [0.0,84.5] [0.0,16.7] [0.0,34596.4] [0.1,1 Iran 50.7 0.0 0.0 4.4 114.6 0.1 1069.2 3.1 (37.5) (0.0) (0.0) (245.5) (0.1) (1782.3) (1.6 [0.0,100.0] [0.0,0.0] [20.4,3071.1] [0.0,0.8] [0.0,14678.0] [0.1,5 Kenya 6.7 0.0 0.0 15.9 1.3 0.3 1140.6 0.5		[0.0,34.2]	[0.0,3.0]	[0.0,17.0]		[0.5,414.0]	[0.0,50.3]	[0.0,159538.5]	[0.0,8.3]
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	indonesia	0.0	0.3	0.0	9.2	6.5	1.0	3451.8	1.3
[0.0,0.0] [0.0,2.0] [0.0,0.0] [0.0,04596.4] [0.1,1,1] Iran 50.7 0.0 0.0 4.4 114.6 0.1 1069.2 3.1 (37.5) (0.0) (0.0) (245.5) (0.1) (1782.3) (1.8 [0.0,100.0] [0.0,0.0] [0.0,0.0] [20.4,3071.1] [0.0,0.8] [0.0,14678.0] [0.1,5 Kenya 6.7 0.0 0.0 15.9 1.3 0.3 1140.6 0.8		(0.0)	(0.5)	(0.0)		(5.7)	(2.7)	(5/62.2)	(0.7)
Instruction Job Columbia O.0 O.0 O.1 1009.2 3. (37.5) (0.0) (0.0) (245.5) (0.1) (1782.3) (1.4) [0.0,100.0] [0.0,0.0] [0.0,0.0] [20.4,3071.1] [0.0,0.8] [0.0,14678.0] [0.1, Kenya 6.7 0.0 0.0 15.9 1.3 0.3 1140.6 0.6	Iran	[0.0,0.0]	[0.0,2.0]	[0.0,0.0]	A A	[U.U,84.5]	[U.U,16.7]	[U.U,34596.4]	[U.1,3.3] 2 4
[0.0,100.0] [0.0,0.0] [0.0,0.0] [20.4,3071.1] [0.0,0.8] [0.0,14678.0] [0.1, Kenya 6.7 0.0 0.0 15.9 1.3 0.3 1140.6 0.5	nail	(37 5)	(0.0)	(0.0)	4.4	(245 5)	(0.1)	1009.2 (1782 3)	.। (1 ৪\
Kenya 6.7 0.0 0.0 15.9 1.3 0.3 1140.6 0.1		[0 0 100 0]	[0 0 0 0]	[0.0,0]		[20 4 3071 1]	0.1)	[0 0 14678 0]	[0 1 9 6]
	Kenva	67	0.0	0.0	15.9	1.3	0.3	114076.0J	[0.1,9.0] () 8
(10.6) (0.0) (1.2) (0.8) (1517.9) (0.8)	onya	(10.6)	(0 0)	(0 0)	10.0	(1 2)	(0.8)	(1517.9)	(0.8)
		[0.0.38.7]	[0.0.0]	[0.0.0]		[0.1.7.2]	[0.0.5.8]	[0.0.6205.0]	[0,1.3.6]
Madagascar 0.0 0.0 0.0 39.3 0.6 0.0 819.1 1.1	Madagascar	0.0	0.0	0.0	39.3	0.6	0.0	819.1	1.2
(0.0) (0.0) (0.3) (0.0) (1036.3) (0.7)	0	(0.0)	(0.0)	(0.0)		(0.3)	(0.0)	(1036.3)	(0.2)

Table A1. Control variables by country

	[0.0,0.0]	[0.0,0.0]	[0.0,0.0]		[0.0,1.8]	[0.0,0.1]	[0.0,3396.6]	[0.5,1.8]
Malawi	0.0	0.0	0.0	17.5	1.4	0.2	237.0	1.1
	(0.0)	(0.0)	(0.0)		(0.4)	(0.1)	(408.5) [0 0 1652 5]	(0.5) [0.4.3.3]
Malavsia	0.0	0.4	0.0	11.7	34.5	0.6	1183.4	0.9
	(0.0)	(0.6)	(0.0)		(14.7)	(1.1)	(1458.0)	(0.5)
	[0.0,0.0]	[0.0,2.0]	[0.0,0.0]		[3.2,57.5]	[0.0,4.3]	[0.0,8094.1]	[0.1,2.0]
Mexico	0.0	0.0	0.4	5.8	59.7	0.4	4188.2	2.0
	(0.1)	(0.0)	(0.8)		(25.0) [6.9.123.5]	(1.5)	(6208.3) [0.0.60164.3]	(1.1)
Mozambique	0.0	0.0	0.0	34.3	2.3	0.2	1147.5	0.5
	(0.0)	(0.0)	(0.0)		(3.6)	(0.5)	(1045.3)	(0.2)
	[0.0,0.1]	[0.0,0.0]	[0.0,0.0]		[0.0,22.1]	[0.0,3.0]	[0.0,3897.9]	[0.0,1.1]
Myanmar	0.0	0.0	0.0	5.5	2.7	0.1	950.5	1.7
	(0.0)	[0.0]	[0.0]		(2.4) [0 1 15 0]	(0.2) [0.0.09]	[0 0 6119 0]	(0.8)
Nepal	1.5	0.0	0.0	59.0	1.7	0.2	549.9	5.2
	(2.1)	(0.0)	(0.0)		(0.8)	(0.1)	(524.5)	(1.2)
	[0.2,12.2]	[0.0,0.0]	[0.0,0.0]	40.0	[0.0,3.3]	[0.0,0.4]	[0.0,1572.7]	[1.8,7.7]
Nicaragua	0.0	0.0	0.0	42.3	40.0	0.1	349.4	0.9
	[0.0]	[0.0,0]	[0.0]		[0 5 231 2]	[0.3]	[0 0 1012 6]	[0.4]
Niger	47.4	0.0	0.0	34.4	2.2	0.1	411.2	0.2
•	(27.3)	(0.0)	(0.0)		(2.1)	(0.5)	(511.0)	(0.1)
NP and a	[0.2,91.9]	[0.0,0.0]	[0.0,0.0]	1.0	[0.3,7.2]	[0.0,4.0]	[0.0,1536.0]	[0.0,0.5]
Nigeria	0.2	0.1	0.0	1.9	4.7	1.1	3870.6	0.3
	[0.0.1.8]	[0.0.1.0]	[0.0.0]		[0.2.58.7]	[0.1.36.6]	[0.0.32142.3]	[0.0.1.1]
Pakistan	31.4	0.0	0.6	3.2	10.8	1.1	9128.8	2.6
	(26.4)	(0.0)	(0.8)		(5.9)	(2.2)	(10789.9)	(2.8)
D	[0.0,98.5]	[0.0,0.0]	[0.0,4.0]	10.5	[0.0,27.5]	[0.0,8.5]	[0.0,45977.1]	[0.0,9.5]
Panama	0.4	0.0	0.0	43.5	19.6	0.1	493.1	1.6
	[0.0.2.1]	[0.0.0]	[0.0.0]		[4.1.34.2]	[0.0.6]	[0.0.1673.1]	[0.6.3.3]
Papua N. Guinea	0.0	0.0	0.0	3.7	0.5	0.9	12.3	1.9
	(0.0)	(0.0)	(0.0)		(0.5)	(2.2)	(63.2)	(0.8)
D	[0.0,0.2]	[0.0,0.0]	[0.0,0.0]	00.4	[0.0,2.1]	[0.0,11.1]	[0.0,338.1]	[0.5,3.4]
Peru	12.1	0.0	0.0	22.1	19.0	0.1	(3776.2)	2.1
	[0.0.63.7]	[0.0.0]	[0.0.0]		[3.1.57.1]	[0.0.3.2]	[0.0.13830.8]	[0.0.4.6]
Philippines	0.0	0.3	0.0	19.8	4.1	1.1	3684.9	1.9
	(0.0)	(0.5)	(0.0)		(1.9)	(3.8)	(5375.3)	(0.6)
Demerie	[0.0,0.0]	[0.0,2.0]	[0.0,0.0]	40.4	[0.3,10.3]	[0.0,22.8]	[0.0,25231.8]	[0.1,5.1]
Romania	(0.2)	0.0	0.4	43.4	85.7 (17.3)	0.3	(1063.3)	1.2
	[0.0.1.3]	[0.0.0]	[0.0.2.0]		[12.3.136.2]	[0.0.7.4]	[0.0.2865.0]	[0.1.3.6]
South Africa	0.3	0.7	0.4	19.0	39.1	0.4	3871.5	2.4
	(0.9)	(0.5)	(0.8)		(15.4)	(0.6)	(3889.8)	(0.9)
Sri Lonko	[0.0,6.6]	[0.0,2.0]	[0.0,2.0]	47.0	[10.2,85.4]	[0.0,2.8]	[0.0,17145.7]	[0.3,4.6]
SITLATIKA	(0.2)	(0.0)	(0,0)	47.9	(3.2)	(0.7)	(302.2)	(0.5)
	[0.0,1.9]	[0.0,0.0]	[0.0,0.0]		[3.1,19.8]	[0.1,3.6]	[0.0,638.9]	[0.0,2.7]
Sudan	35.3	0.0	0.0	33.3	4.6	0.0	1583.8	0.4
	(32.3)	(0.0)	(0.0)		(5.8)	(0.1)	(2200.4)	(0.3)
Tajikistan	[0.0,99.5]	[0.0,0.0]	[0.0,0.0]	26.3	[0.0,34.1]	[0.0,0.3]	275.1	[0.0,1.5]
Гајткізтан	(17.9)	(0.0)	(0.0)	20.5	(12.2)	(1.2)	(342.6)	(2.4)
	[0.4,66.9]	[0.0,0.0]	[0.0,0.0]		[0.0,68.5]	[0.0,4.0]	[0.0,832.0]	[0.8,9.3]
Tanzania	0.0	0.0	0.0	21.4	1.6	0.4	862.7	0.6
	(0.1)	(0.0)	(0.0)		(1.8)	(0.8)	(1597.7)	(0.4)
Thailand	[0.0,0.9]	[0.0,0.0]	[0.0,0.0]	6.1	[0.0,6.6] 24.3	[0.0,3.0]	[0.0,6933.2] 1154.4	[0.0,1.6] 1.2
manana	(0.0)	(0.2)	(0.0)	0.1	(10.5)	(0.7)	(1932.0)	(0.6)
	[0.0,0.1]	[0.0,1.0]	[0.0,0.0]		[4.1,81.0]	[0.0,3.7]	[0.0,9905.9]	[0.0,2.9]
Turkey	0.5	0.3	0.0	9.1	53.9	0.3	2326.3	2.8
	(0.7)	(0.6)	(0.0)		(52.3)	(0.6)	(5007.4)	(1.2)
Uganda	0.3	[0.0,3.0]	0.0	14.8	0.4	0.8	270.5	[0.3,0.0] 1 2
- 3	(0.9)	(0.0)	(0.0)		(0.8)	(2.2)	(667.9)	(0.7)
	[0.0,5.4]	[0.0,0.0]	[0.0,0.0]		[0.0,4.1]	[0.0,11.0]	[0.0,2577.0]	[0.2,3.9]
Venezuela	0.0	0.1	0.0	43.6	57.6	0.2	2891.9	1.5
	(0.1)	(0.3)	(0.0)		(28.4)	(0.3)	(3106.4)	(1.1)
Vietnam	[0.0,0.2] 0.1	0.0	0.0	9.0	6.2	0.4	[0.0,12103.0] 778.3	[0.0,4.2] 2.2
	(0.1)	(0.0)	(0.0)	5.0	(2.5)	(0.8)	(1696.0)	(1.2)
	[0.0,0.6]	[0.0,0.0]	[0.0,0.0]		[0.5,14.3]	[0.0,9.3]	[0.0,11118.1]	[0.0,6.1]
Full sample	6.2	0.1	0.3	16.9	21.0	0.6	7627.2	1.7
	(17.2)	(0.4)	(1.5)		(51.6)	(2.2)	(21659.8)	(1.5)
	[0.0,100.0]	[0.0,6.0]	[0.0,22.0]		[0.0,3071.1]	[0.0,50.3]	[0.0,481157.6]	[0.0,9.6]

Notes: The table displays means, (standard deviations) and [min,max] for all control variables used in the analysis. For the binary indicator "Capital city" only percentages are reported. ^a Raw night light intensity values can reach a maximum of 63. Night light intensity per 1,000 people was calculated by first aggregating night light emissions across grid cells within each ADM region and by dividing by the population (times 1,000) living in the respective area. Afterwards, for each disaster, area-weighted averages were created. ^b For readability purposes, population density is scaled in terms of 1,000 people per km² and city population is scaled in terms of thousands of inhabitants. ^c Ruggedness is scaled in terms of 100s of meters of elevation differences.